

Tectonics

RESEARCH ARTICLE

10.1029/2019TC005965

Key Points:

- Calcareous sediments seaward of the Hikurangi Margin host stylolite and fault clusters formed by mixed-mode deformation at <1-km depth
- We compare strains from stylolite mass loss to analytical models; their different magnitudes imply an open system with advective transport
- At depths where slow slip occurs, high calcite concentration and solubility favor high-pressure solution rates and thus frequent stylolites

Supporting Information:

Supporting Information S1

Correspondence to:

H. Leah, LeahHR@cardiff.ac.uk

Citation:

Leah, H. R., Fagereng, Å, Meneghini, F., Morgan, J. K., Savage, H. M., Wang, M., et al. (2020). Mixed brittle and viscous strain localization in pelagic sediments seaward of the Hikurangi Margin, New Zealand. *Tectonics*, *39*, e2019TC005965. https:// doi.org/10.1029/2019TC005965

Received 4 NOV 2019 Accepted 12 JUL 2020 Accepted article online 17 JUL 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Mixed Brittle and Viscous Strain Localization in Pelagic Sediments Seaward of the Hikurangi Margin, New Zealand

Harold Leah¹, Åke Fagereng¹, Francesca Meneghini², Julia K. Morgan³, Heather M. Savage⁴, Maomao Wang⁵, Rebecca Bell⁶, and Matt J. Ikari⁷

¹School of Earth and Ocean Sciences, Cardiff University, Cardiff, UK, ²Dipartimento di Scienze della Terra, Universit degli Studi di Pisa, Pisa, Italy, ³Department of Earth Science, Rice University, Houston, TX, USA, ⁴Department of Earth and Planetary Sciences, University of California, Santa Cruz, CA, USA, ⁵College of Oceanography, Hohai University, Nanjing, China, ⁶Basins Research Group, Imperial College London, London, UK, ⁷MARUM Center for Marine Environmental Sciences and Faculty of Geosciences, University of Bremen, Bremen, Germany

Abstract Calcareous-pelagic input sediments are present at several subduction zones and deform differently to their siliciclastic counterparts. We investigate deformation in calcareous-pelagic sediments drilled ~20 km seaward of the Hikurangi megathrust toe at Site U1520 during International Ocean Discovery Program (IODP) Expeditions 372 and 375. Clusters of normal faults and subhorizontal stylolites in the sediments indicate both brittle faulting and viscous pressure solution operated at <850 m below sea floor. Stylolite frequency and vertical shortening estimated using stylolite mass loss, porosity change, and distribution increase with carbonate content. We then use U1520 borehole data to constrain a P-T-t history for the sediments and apply an experimentally derived pressure solution model to compare with strains calculated from stylolites. Modeled strains fail to replicate stylolite-hosted strain distribution or magnitude, but comparison shows porosity, composition, and grain-scale effects in diffusivity and mass transfer pathway width likely exert a strong influence on pressure solution localization and strain rate. Stylolite and fault clusters concentrate clay in these sediments, creating weak volumes of clay within carbonates, that may localize slip where the plate interface intersects the carbonates at <5-km depth. Plate interface slip character and rheology will be influenced by the deformation of intermixed phyllosilicates and calcite, occurring by variably stable frictional slip and pressure solution of calcite. Pressure solution of calcite is therefore important at the shallow plate interface, waning at the base of the slow-slipping zone because calcite solubility is low at temperatures >150°C where frictional (possibly seismic) slip likely predominates.

Plain Language Summary The type of sediments entering subduction zones will influence the way the plates in the subduction zone slide past one another. We looked at limestones in sediments drilled before they reach the subduction zone and found that because of the pressure they are under, they begin to crack and dissolve at very shallow depths. Most of the dissolution happens on thin layers where it concentrates clay by removing other, more soluble minerals. We compare how much vertical shortening we see in the sediments to a computer model. The model overestimates vertical shortening over the history of the sediment unless either high-pressure fluids reduce the pressure felt by the sediments or dissolution is governed by the largest sediment grains rather than their average size. Dissolving and cracking make the sediments weaker by concentrating soft materials such as clay. When these sediments enter the subduction zone, the two plates might slip past one another more easily on these weak regions, possibly during slow slip events.

1. Introduction

The mechanical behavior of carbonate-rich sediments during subduction is not well understood, despite their recognition at several margins around the world (Moore & Mascle, 1990; Morris et al., 2006; Wallace et al., 2019). Rock deformation experiments have, however, shown that carbonates deform very differently to siliciclastic sand and mudstones typically studied on subduction margins (Boulton et al., 2019; Ikari et al., 2013; Kurzawski et al., 2018; Rabinowitz et al., 2018). For pressure (P) and temperature (T) conditions at



Experimental deformation by pressure solution has been observed and modeled extensively (Macente et al., 2018; Rutter, 1976, 1983; Zhang et al., 2002, 2010), and stylolites resulting from localized pressure solution in natural settings are well characterized (Fletcher & Pollard, 1981; Laronne Ben-Itzhak et al., 2014; Lind, 1993; Peacock & Azzam, 2006; Tondi et al., 2006; Toussaint et al., 2018; Viti et al., 2014). Stylolites form by dissolution of material on discrete surfaces, or in thin tabular volumes, at elevated rates relative to the surrounding rock (Durney, 1972; Toussaint et al., 2018). Therefore, stylolites are recognized by a higher content of relatively insoluble minerals, caused by passive concentration during intergranular pressure solution. Although the importance of pressure solution in the shallow deformation of carbonates is well documented (Baud et al., 2016; Gratier et al., 1999, 2013; Heald, 1955; Renard et al., 2000; Tondi et al., 2006), many of the controlling factors linking pressure solution to distribution, kinetics, and dynamics of stylolite formation remain unclear.

Several factors influence the rate of pressure solution, including clay content (Aharonov & Katsman, 2009; Hickman & Evans, 1995), grain surface area (Gratier et al., 1999), mineral solubility (Rutter, 1976), grain contact area and microstructure (including variations due to porosity and grain boundary healing) (Croizé et al., 2010; Ebner et al., 2010; van den Ende et al., 2019), fluid flow and composition (Lehner, 1995; Zhang & Spiers, 2005), and gradients in normal stress (Rutter, 1976). The constitutive equations for creep rate by pressure solution are distinct when rate limited by either dissolution, diffusion, or precipitation, and each is consequently described by a different model (Gratier et al., 2013; Raj, 1982; Rutter, 1976). Pressure solution on stylolites at shallow depth represents viscous deformation in a depth range typically associated with brittle fracture and frictional sliding (<15 km Paterson & Wong, 2005; Tada & Siever, 1989). At constant environmental conditions, the rate of pressure solution is dominantly controlled by mineral solubility and grain size, facilitating greater ductility of the upper crust in lithologies containing more soluble minerals (Gratier et al., 1999; Renard et al., 2000). The brittle and viscous components of deformation in the upper crust therefore vary in relative intensity with lithology as well as depth and temperature. Where elevated rates of pressure solution form stylolites, changes in properties such as clay content and porosity affect localization and frictional slip properties of faults (Baud et al., 2016; Tondi et al., 2006; Viti et al., 2014; Watkinson & Ward, 2006).

International Ocean Discovery Program (IODP) Expeditions 372 and 375 drilled and sampled ve sites on a transect across the Hikurangi Margin, New Zealand, between November 2017 and May 2018 (Wallace et al., 2019). We study calcareous-pelagic chalks and marls in the incoming sedimentary sequence sampled at Site U5120. We measure fault and stylolite distribution and make microstructural observations to determine the extent and character of pre-subduction deformation before calculating uniaxial strain from pressure solution using chemical mass loss estimates and stylolite frequency. A possible relationship between CaCO₃ content and stylolite frequency could provide an insight into a key natural control on stylolite development.

The abundant data gathered during IODP Expedition 375 provide a unique opportunity to constrain the P-T-t history for the pelagic sequence at Site U1520, which we use to apply published models of intergranular pressure solution (Rutter, 1976, 1983). We then compare these results to those calculated from stylolite frequency earlier in this study and nd agreement requires sustained high fluid pressures or larger length scales of material transport than the average grain size of lithologies hosting stylolites. Finally, we extend this model to the future subduction of the sediments and discuss the rheological impact of ongoing pressure solution and pre-subduction strain weakening on current subduction.

2. The Hikurangi Margin

The Hikurangi Margin subduction zone dips northwest beneath the east coast of New Zealand's North Island (Figure 1a). It accommodates westward subduction of the Hikurangi Plateau beneath the Australian plate at a rate varying along strike from 40 to 50 mm year⁻¹ (Wallace, 2004). Slow slip events





Figure 1. Hikurangi Margin map and section and stratigraphic logs from Site U1520, cored during IODP Expedition 375. (a) Map shows contours of the plate interface depth (black) and slip during the October 2014 slow slip event (white) characterized by Wallace et al. (2016) east of North Island, New Zealand (inset map). The red point is IODP Expedition 375 drill Site U1520, on the white line showing the 05CM-04 seismic section drawn below (c). The Hikurangi Trench is in red. (b) Lithostratigraphic log shows core from site U1520. Colors of units (yellow, green, brown, and blue), and contacts between units (orange, green, and brown) are the same on section (c) below. Schematic section (c) is redrawn from Barnes et al. (2020) and shows the relationship of the units cored at U1520 to the plate interface. Also shown are the décollement (thick black line), active faults (red lines), inactive faults (dark red lines), normal faults (thin black lines), base of volcanic basement (light blue line), approximate slow slipping zone, and the high-reflectivity zone (HRZ) as characterized by Bell et al. (2010).

on the northern Hikurangi Margin accommodate up to 250 mm of slip over several days or weeks and recur every 2 5 years (Wallace & Beavan, 2010; Wallace et al., 2016). As well as slow slip, the northern part of the Hikurangi Margin is characterized by episodic seismicity of variable character, including large tsunami earthquakes (M_w 7.2, Clark et al., 2019; Doser & Webb, 2003; Wallace et al., 2014), microseismicity (Delahaye et al., 2009), and tectonic tremor (Todd & Schwartz, 2016; Yabe et al., 2014).

Seismicity at the Hikurangi Margin is distributed throughout the sedimentary upper plate and oceanic lower plate, with tsunamigenic earthquakes occurring on the plate interface (Clark et al., 2019; Shaddox & Schwartz, 2019; Todd et al., 2018). The volume hosting slow slip likely encompasses the plate interface and shallow earthquake hypocenters, extending from near the trench to 15- to 20-km depth (Wallace et al., 2016), possibly hosted in heterogeneous lithologies characterized on the incoming plate (Barnes et al., 2020; Wallace et al., 2019). Seismic transects across the margin have shown relatively thin sediment cover on the incoming plate (1 km) between large seamounts (Barker et al., 2009; Davy et al., 2008). High-amplitude reflections are present in both the incoming plate and downdip of the slow slip-hosting volume,



suggesting incoming material may be underthrust (Bell et al., 2010). The thin sediment cover, moderate convergence rate, rough incoming topography, and steep taper angle at this margin suggest it may be undergoing frontal erosion and is consequently likely to underthrust sediment rather than accrete it (Barker et al., 2009; Fagereng, 2011b).

3. Methods

3.1. IODP Drilling at Site U1520

Site U1520 is on the flank of the T \bar{u} ranganui Knoll seamount and therefore represents a slightly condensed incoming sedimentary sequence. Logging-while-drilling (LWD) data were collected at Site U1520 during Expedition 372 (Wallace et al., 2019). The same site was subsequently cored and wireline logged during Expedition 375, from 0 to 642.3 m below sea floor (mbsf) in hole U1520D, and from 646.0 to 1054.1 mbsf in hole U1520C (Wallace et al., 2019). Cores revealed a ~1-km-thick sequence of heterogeneous sediments including turbiditic silts and muds (Units I III), pelagic muds and carbonates (Unit IV), and volcaniclastic conglomerates with minor mudstones (Units V and VI; Figure 1b). In this work, we are concerned with the pelagic carbonates and carbonaceous muds which comprise Unit IV. Unit IV was found at depths of 509.82 848.45 mbsf (Wallace et al., 2019).

Site U1520 is approximately 20 km seaward of the toe of the Hikurangi megathrust (Figure 1a). Assuming a convergence rate of ~40 mm year⁻¹, the sediments intersected at Site U1520 will reach the toe of the thrust in approximately 0.5 Myr. Comparing the depth of Unit IV at Site U1520 with interpretations of active seismic data along line 05CM-04 (Barker et al., 2018), the shallower and deeper contacts likely correspond to the bases of seismic Units 5 and 8 (Barnes et al., 2020). These same horizons can be traced landward until truncated by the megathrust at depths of ~4.5 and ~5 km (Figure 1c), where the mud- and chalk-dominated rocks are inferred as a key protolith interval in which the plate interface will likely locate (Barnes et al., 2020).

Measurements of many physical and chemical properties were collected by LWD and shipboard core logging, details of which are described by Wallace et al. (2019). Here, we use data concerning the density, temperature, age, porosity, mineralogy, and lithology of sediments in Unit IV of Site U1520.

Relative mineral abundance from X-ray diffraction (XRD) is normalized using coulometric $CaCO_3$ content analysis (Wallace et al., 2019). Relative abundances of minerals determined by XRD were calculated for a simpli ed suite of four mineral groups: calcite, quartz, feldspars, and clays. Clays are measured as a bulk single mineral, though dominant species present in Unit IV are smectite and illite (see XRD methods in Wallace et al., 2019).

3.2. Conditions of Pre-Subduction Compaction on Stylolites

The physical properties of these sediments were logged by LWD during Expedition 372 and on extracted core during Expedition 375. From shipboard density measurements (i.e., moisture and density in Wallace et al., 2019), in situ effective vertical stress (σ_v^{eff}) was calculated as

$$\sigma_v^{eff} = g_{i=1}^n (\rho_{bi} - \rho_w) dz_i \tag{1}$$

where there are *n* intervals of bulk density measurement spacing dz with measured bulk density ρ_b . Gravitational acceleration (g) is equal to 9.81 ms⁻¹, and average seawater density (ρ_w) is assumed to be 1,024 kg m⁻³ (Nayar et al., 2016). Typical values for $dz_{(i)}$ are ~1 m, though this varies with gaps in coring and core recovery. Hydrostatic (and, for comparison, near-lithostatic) pore fluid pressure is assumed throughout the studied depths. The results of the hydrostatic pore fluid pressure calculation (Figures 2a and 2b) show that effective stress in Unit IV increases approximately linearly from ~4 MPa near the top (509.8 mbsf) to ~8 MPa near the base of the unit (848.5 mbsf).

Five of the seven downhole temperature measurements conducted in the shallowest 250 m of hole U1520D form a linear trend corresponding to 38° C km⁻¹ (Figure 2c). The other two values appear as outliers and are therefore ignored. Extrapolating a constant trend from these measurements to depths greater than the shallowest 250 mbsf, temperature increases from 19.4°C at 509.8 mbsf to 32.2°C at 848.5 mbsf (Figure 2c). A







Figure 2. Physical conditions with depth in Site U1520 Units I IV (contact depths shown as gray lines). Plots show (a) density, (b) effective vertical and hydrostatic stress, (c) temperature, and (d) age measurements from shipboard analysis during IODP Expedition 375. Note that hydrostatic stress in (b) starts at \sim 35 MPa, corresponding to the water depth of 3522.1 m.

detailed discussion of geothermal gradient in this type of sediment is beyond the scope of this work, so we will focus on the conditions under which strain was likely accommodated in Unit IV. To constrain the depositional age of the sediments at Site U1520, we use an approximate age model based on nannofossil, planktonic foraminifera, and palaeomagnetic reversal ages (Figure 2d Wallace et al., 2019).

3.3. Characterising Stylolite and Fault Frequency and Texture

To determine pre-subduction mechanisms of sediment deformation, we counted faults and stylolites in core samples of lithostratigraphic Unit IV. Image scans of drill core (collected shipboard using the Section Half Imaging Logger, see core handling and analysis methods in Wallace et al., 2019) were used to visually identify stylolites and faults (e.g., Figure 3a). For the purposes of counting, stylolites were de ned as typically dark, slightly wavy surfaces where material loss was apparent and concentration of insoluble materials was inferred. Faults were de ned as closed or recrystallized fractures with visible apparent offset. The use of these de nitions aided in discounting ne clay-rich sedimentary layers and drilling damage from stylolite and fault populations. Horizons of increased deformation measured here coincide with those noted by shipboard structural geologists based on direct observation of the core (Wallace et al., 2019). The depth where a fault or stylolite occurred within the core was recorded as the middle of an interval de ned by the intersection between the planar structure and the two sides of the core liner. As noted by shipboard scientists measuring selected structures in 3D (Wallace et al., 2019), most stylolites are approximately horizontal.

Six samples were collected from throughout the pelagic carbonate sequence (see repository les for data collected from all samples) and set in epoxy resin to avoid breakage during thin section preparation. By careful cutting and grinding using silicon carbide grit slurry and a Logitech PM5 polisher with 0.3μ m aluminum oxide, polished thin sections were made from samples cut either perpendicular to bedding or parallel to lineation and perpendicular to fault surfaces. Polished thin sections were used to produce optical photomicrographs, backscatter electron (BSE) images and energy dispersive spectra (EDS). BSE and EDS data were captured using a Zeiss Sigma HD Field Emission Gun Analytical scanning electron microscope (SEM) at the School of Earth and Ocean Sciences in Cardiff University. BSE images were collected at 20 keV with an aperture of 120 μ m and a working distance of 8.9 mm. We use the number of dark pixels per row of BSE image as a proxy for porosity as BSE returns no result when analyzing a void. Following Verberne and Spiers (2017) and Heilbronner and Barrett (2014), images were cropped, rotated, histogram matched using the •Match



Figure 3. Texture of stylolites in Unit IV of the U1520 input sequence. Stylolites show (a) faint layering de ned by darker layers, (b) variably-diffuse clay-rich layers at the microscale, (c) darker en-echelon seams anastomosing around burrows in clay-rich material and dying out in clay-poor, fossiliferous regions. SEM BSE images show (d) fossils dissolved on stylolite surfaces, (e) ne grained ($<10 \mu$ m) clay within a stylolite seam. SEM EDS maps show Ca depletion and Si, Mg, Al, and Fe enrichment within stylolite seams from (f and g). Colors in EDS maps are blended, so lighter colors between stylolite seams represent the co-occurrence of several elements.

Color function in Adobe Photoshop, and Gaussian Itered with a Gaussian width of 4 before the algorithm of Otsu (1979) was applied to determine a darkness threshold between 0 and 255 corresponding to porosity. Pixels values below this value (52 for sample 20R4W and 50 for sample 19R1W) were counted, and their ratio to the total number of pixels per row was averaged over approximate areas to show a bulk porosity change. Where cracks were present, we cropped the averaging area; both cropped and uncropped averaging areas within the stylolite are shown in Figure 4b. For EDS mapping of faults and stylolites, spot size was either 1.59, 2.49, or $2.66 \,\mu$ m, accelerating voltage was 20 keV, dwell time was 100 μ s, and aperture size was 120 μ m. For determination of element concentrations from SEM EDS images, relative weight percentages were calculated in Oxford Instruments AZTEC software. The element concentrations calculated are relative within a single mapped area.

4. Description of Stylolites and Faults

Macroscopically, more well-developed stylolites appear as composite structures comprising discrete dark seams, frequently linked with more subtle seams within the adjacent few centimeters of core. Some stylolitic surfaces within Unit IV have relatively steep apparent dips (>45°) and host minor shear offset on undulating dark surfaces. These possibly represent poorly developed early fault surfaces which have later undergone pressure solution or •healing (Figure 3a). At the microscale, most seams are composed of several ner seams (Figures 3b and 3c), with their most intense shared anastomosing branch representing the macroscopically visible seam. Seams within these ner networks are relatively thin (<0.1 mm), undulating, and sub-parallel to bedding (Figure 3b). Calcareous microfossils of uniform size are common throughout





Figure 4. Porosity across variably-developed stylolites from SEM BSE images. Both the less well developed stylolite seam (left) and more well developed stylolite seam (right) show less darker shades and visibly lower porosity within the stylolite seam. SEM BSE images are rotated so the stylolite seam is parallel, and histogram normalized to distribute the image across the entire range of gray (0 is black, 255 is white), and the fraction of dark pixels (<52 or <50) per row is used as a porosity measurement.

carbonaceous stylolite-bearing lithologies and are frequently dissolved on seams throughout Unit IV. Fossiliferous horizons coincide with thinner, less pervasive stylolites when compared to less fossiliferous, more clay rich horizons (Figure 3c). Burrows have a randomized internal structure, within which discrete stylolitic horizons are not visible (Figure 3c). In contrast, the margins of burrows are more concentrated in clay-rich material, and stylolites there are clearly visible and well-developed.

SEM images of stylolites show dissolved edges of porosity-hosting microfossils and concentration of ne clays on stylolite seams (Figures 3d and 3e). Microfossils up to 200 μ m are the largest grains within the chalk and clearly show dissolution on thin (<20 μ m wide) stylolite seams (Figure 3d). The seams themselves concentrate along strings of adjacent fossils and host very ne clays (<5 μ m) and have in places almost entirely dissolved fossils, resulting in collapse of their interior porosity (Figure 3e). Remnant material within stylolite seams shows a preferred subhorizontal orientation, in places mirrored in the matrix directly adjacent to the seam (Figures 3d and 3e).

Element maps show a Ca-rich protolith matrix with stylolite seams relatively enriched in magnesium, silicon, iron, and aluminum (Figures 3f and 3g). The bulk stylolite hosts increased silicon and magnesium while individual seams appear to dominantly concentrate iron (Figures 3f and 3g). Individual seams are of variable thickness and anastomose throughout the width of the stylolite to form a weakly connected network with scattered parallel fractures (Figures 3f and 3g). Fractures are open but host some aluminum, probably representing clays seen within seams in BSE images (Figures 3d and 3e). Open fractures could be a result of sample preparation, where clay was removed by polishing, or core extension, where extension parallel to the core axis occurs when retrieving core from depth. Because we interpret them as formed either by sample preparation or by drilling-induced damage, these open fractures are not included in any discussion or quanti cation of subseafloor deformation.

From BSE images and EDS maps, it is qualitatively clear that stylolites have lower porosity than their surroundings (Figures 3d 3g); mean measured porosities adjacent to two stylolites are between 0.41 and 0.77 (Figure 4). Shipboard porosity measurements on bulk intact samples from within 1-m (19R1W) and 7-m (20R4W) depth of the samples shown in Figure 4 were between 0.36 and 0.42, below our estimates from image analysis (Figure 2 Wallace et al., 2019). Porosity measurements within well-developed stylolites are complicated by seam-parallel open fractures which host large areas of pore space (Figure 4), and we therefore analyze well-developed stylolites both with and without open fractures. Mean porosities within





Figure 5. Texture of faults in Unit IV of the U1520 input sequence. Images show (a) localized sharp fault surface formed adjacent to clay-rich regions in ne-grained marl, (b) optical photomicrograph of the area shown in (a), (c) enlargement of rare veins in footwall of fault, area shown in (b), and (d) optical photomicrograph of footwall fracture hosting offset. SEM BSE image shows (e) common fractures with offset hosting brighter ne grained material, and EDS map (f) details ne-grained material around fractures comprised of dominantly aluminosilicate minerals (orange).

stylolites vary from 0.08 to 0.26, corresponding to reductions of 0.35 0.42 compared to the surrounding sediments and ignoring fractures (Figure 4). With further dissolution to develop a stylolite further, porosity loss is likely to increase (Toussaint et al., 2018), suggesting that 0.4 is a more appropriate gure for mean porosity loss in moderately well-developed stylolites. This porosity loss is a mean change over anastomosing seams and interseam areas of the bulk stylolite, commonly hosting higher porosity (0.4 0.8) where undissolved shells are present and lower porosity (<0.2) where clays are concentrated (Figure 4). Because pore-space within intact shells becomes connected to other porosity as soon as shells breakdown either mechanically or by dissolution, we include this fossil-hosted porosity in our measurements.

Faults form localized slip surfaces within centimeter-scale fracture zones (Figures 5a and 5b). Slip surfaces are sharp and smooth clay surfaces (Figure 5a), frequently with well-de ned straight striae showing slip direction. Due to their formation within a sedimentary stack under uniaxial vertical shortening, faults are assumed to be normal, con rmed by a normal sense of shear reported wherever shear sense was obtained from stepped, striated fault surfaces (Wallace et al., 2019). Apparent displacements on faults range from a few millimeters to greater than the observed length of the fault (12 cm, depending on apparent dip). Fault surfaces host little mineralization (Figures 5b and 5d), with striae likely forming by frictional wear. Fractures within the centimeter-scale damage zones around fault surfaces are mostly a combination of extension and shear fractures and are mostly open, rarely •healed by veining (Figures 5b 5d). Clay-rich material is present as relatively thick (3 mm) seams in the immediate vicinity of some slip surfaces (Figures 5a 5d). In SEM BSE images, it is clear that thicker clay-rich seams adjacent to faults are connected to similar, thinner, seams which run along fractures adjacent to the fault surface (Figure 5e). EDS maps show enrichment of magnesium, iron, silicon, and aluminum throughout the clay-rich network, including the material comprising the fault surface (Figure 5f).

Faults and stylolites rarely occur on the same surface within Unit IV, except where early faults have been exploited by pressure solution (Figure 3a). Where faults and stylolites occur within the same depth range, stylolites are typically intersected and offset by faults. Some faults host clay-rich material immediately

Tectonics





Figure 6. Summary of lithology, mineralogy, porosity, and stylolite and fault frequency throughout Unit IV. Plots show (a) simpli ed stratigraphy, (b) mineralogy, and (c) porosity from shipboard data (Wallace et al., 2019). Depth-frequency plots of (d) stylolites and (e) faults (red) are from observations of core image scans. Both (d) stylolite and (e) fault plots show cumulative frequency (blue and red ll, respectively), frequency per meter (white ll). Calcium carbonate content is also shown on stylolite plot (gray lled line in d).

adjacent to their slip surface, possibly indicating the existence of stylolites prior to faulting. The existence of stylolites adjacent to fault slip surfaces is uncertain as faults obscure the stylolite texture. These possible stylolites have therefore been excluded from stylolite frequency counts.

5. Distribution of Faults and Stylolites in Unit IV

5.1. Downhole Frequency of Stylolites and Faults

Following the evidence for pre-subduction deformation on faults and stylolites, we now quantify their frequency and distribution within core samples. To analyze the distribution of structures as a function of lithology, we use a simpli ed lithostratigraphic sequence, ignoring minor lithologic intervals (<5 m thick) to get seven lithological units (Figure 6).

Stylolites in Unit IV occur in distinct clusters within similar grain size sediments, increasing in intensity with depth. Clusters are more frequent and intense where CaCO₃ content is >50 wt%, most notably in chalk near the deeper contact with the volcaniclastic conglomerate (Figure 6). The deepest chalk unit sits between 797.99 and 848.45 mbsf, contains >90-wt% CaCO₃, and hosts ~75% of stylolites in the section. Other signi - cant peaks in stylolite frequency are in the two marl units (655.15 720.93 and 738.68 773.26 mbsf). Minor peaks in stylolite frequency occur throughout the deeper part of the sequence in the marl, chalk, and debris flow units (Figure 6). Stylolites are very rare in the calcareous mudstone and conspicuously absent in the mudstone unit, which contains less CaCO₃ (<50 wt%). Similar to stylolites, faults are also clustered, but fault clusters are more distributed throughout the core than stylolite clusters.

5.2. Clustering Analysis of Stylolites and Faults

The coef cient of variation (C_{ν}) is calculated as the ratio of the standard deviation of distances between features to the mean of distances between features (Gillespie et al., 1999)





Figure 7. Normalized length-spacing coef cient of variation analysis for stylolites (left) and faults (right). Coef cient of variation was calculated using incrementally thresholded spacings, then normalized to lithology thickness. Points indicate 1 m threshold thickness for each unit.

$$C_v = \frac{SD(s)}{s} \tag{2}$$

where SD(s) represents the standard deviation of the sample of distances between each fault or stylolite and *s* represents the mean of the same sample distances. A C_{ν} of 0 represents a periodic distribution, whereas a C_{ν} of 1 represents a perfectly random distribution, and a C_{ν} greater than one represents clustered features (Cox & Lewis, 1966). Coef cient of variation analysis has been carried out on one-dimensional transects of discrete features to show the self-arrangement of stylolites (Railsback, 1998) and veins (Fagereng, 2011a; Gillespie et al., 1999). Following the approach of Gillespie et al. (1999), we threshold the distances between features at incrementally larger values to discern over what length-scale clustering is apparent. To compare stylolite and fault distributions within each lithological unit, we present these data as C_{ν} versus threshold thickness normalized to the thickness of the simpli ed lithological unit (Figure 7).

Whole sample C_{ν} values for faults are smaller than for stylolites throughout the studied threshold range, representing less intense clustering of faults than stylolites (Figure 7). This is also reflected in the unthresholded single value analysis ($C_{\nu}^{\text{stylolite}} = 6.993$, $C_{\nu}^{\text{fault}} = 4.426$). Stylolite spacing C_{ν} -threshold scaling relationships appear to follow a power-law relationship within Unit IV (Figure 7; cf. Gillespie et al., 1999). The degree of clustering of stylolites increases with smaller spacing in almost all lithologies; stylolites spaced >1 m are randomly to periodically distributed in all lithologies (Figure 7). The periodic nature of stylolites spaced at thresholds >10 m probably reflects edge effects of the 350-m section of core studied, limiting analysis over this scale. This is also shown by low C_{ν} values in the shallowest calcareous mudstone, where only seven stylolites are present over a short section of core, compared to the entirety of Unit IV, where 873 stylolites are present (Figures 6 and 7). Stylolites are most clustered when spaced <1 cm apart in the deepest chalk unit. In all other considered lithologies, stylolites spaced 1 cm to 1 m are similarly clustered regardless of lithology (Figure 7).

Fault spacing C_v -threshold scaling relationships are more variable between lithologies than stylolites, though faults still cluster at spacings <1 m in randomly distributed horizons throughout Unit IV (Figures 6 and 7). Clustering within the marl, calcareous mudstone, and chalk lithologies is similar over all studied scales. Almost all faults spaced less than ~1 m in these lithologies have $C_v > 1$, indicating clustering. When faults are ~1 m apart, C_v is ~1, indicating a random distribution (Figure 7). Faults in the debris flow are consistently more clustered than in other lithologies, maintaining a higher C_v throughout almost the entire threshold range. In stark contrast, the 12 faults within the mudstone are periodic throughout the studied threshold range (Figure 7).



6. Strain Within Unit IV

Strain within Unit IV of Site U1520 was accommodated by mixed brittle and viscous deformation in the form of faults and stylolites. At sites of stylolite and fault co-occurrence, stylolites are mostly cut by fault surfaces, suggesting faulting occurred either more recently or more rapidly than stylolite growth. Faults and stylolites have therefore occurred over distinct time scales; stylolites appear to have accommodated strain more gradually through the history of the sediment while faulting occurred as a transient process, accommodating displacement rapidly before ceasing (Gratier et al., 1999). Unit IV can be traced landward and downdip to the active plate interface (Figure 1). With this in mind, we consider how vertical uniaxial strain from sedimentary load has been accommodated in Unit IV prior to subduction and describe how clustered faults and stylolites affect the initial rheology of sediments which could be accommodating strain at the plate interface.

6.1. Mass and Volume Loss Within Stylolites

Isocon plots use a known immobile species or, as in this case, an element present in a relatively insoluble mineral, to quantify mass changes between comparable analyses. Aluminum is present in clays and other relatively insoluble minerals (Aharonov & Katsman, 2009; Gratier et al., 2015; Hickman & Evans, 1995), commonly visible around and within calcareous fossils in the nearby protolith (Figures 3d 3g). Original depositional clay content of the analyzed areas within and without the stylolite is likely different as stylolites are more likely to form in areas of elevated clay concentration (Toussaint et al., 2018). The self-organizing nature of pressure solution, however, is likely to amplify these original differences as insoluble minerals are passively concentrated in zones of dissolution as more soluble carbonate is removed (Heald, 1955). We calculate bulk mass change by plotting the isocon through the plot origin and aluminum (Figure 8, Grant, 1986) and note that this method will overestimate mass loss if the stylolite formed in an area of initial high clay content. Titanium is also commonly used in isocon analyses, but low concentrations in our samples yield large errors and titanium was therefore not used here. From the bulk mass change, relative changes in calcium wt% are calculated (Figure 8).

Despite minor plagioclase (<10 wt%) throughout Unit IV (Figure 6), none was found in our EDS maps, reiterating that calcite is the dominant calcium-hosting phase losing mass on stylolites in originally calcite-rich sediments. The loss of calcite is consistent with (1) dissolution of fossils (Figures 3d 3g), (2) clusters of stylolites forming peaks of increased frequency with high $CaCO_3$ content (Figures 6 and 7), and (3) the relatively higher solubility of calcite, compared to silicates, under the studied conditions (Heald, 1955; Plummer & Busenberg, 1982), showing these data are consistent with mass loss by dissolution in calcareous rocks. We cannot constrain the amount of CO_2 lost from calcite in our samples. The bulk calcium mass change is therefore used to calculate volume change using calcite density and porosity.

Mass change in stylolite seams relative to the wall rock varies but is generally dominated by loss of Ca and gain of Mg, Si, Al, Ti, K, and Fe (Figure 8). Bulk mass change calculated within stylolite seams shows loss of Ca-rich material, probably calcite, and concentration of aluminosilicates in highly localized zones (Figure 8). Relative total and calcium mass change estimates presented here are -11% to -45% and -3% to 49%, respectively (Figure 8). Despite the possibility of overestimating mass loss due to initially higher concentrations of clay in stylolite-forming volumes, these mass losses are consistent with previous estimates of bulk mass change of -16% to -44% and CaCO₃ change of -15% to -45% from early pressure solution in marl from Lacroix et al. (2015).

Relative elemental mass changes in stylolites are variable between individual seams, but the majority of strain derives from porosity loss within the seam due to closer grain packing and ne aluminosilicates lling in intergranular voids (Figure 3). There is, again, the possibility that some of this calculated porosity change reflects initial spatial differences. We may overestimate the temporal volume change if stylolites formed on clay-rich, low-porosity horizons, or underestimate it if stylolites reflect initial high-porosity horizons where local normal stresses were elevated. We assume a porosity change of -40% based on porosity calculation from SEM BSE images (Figure 4), producing volume losses of 56% and 67 73% for bulk stylolites and individual seams, respectively (Figure 8). Our calculated bulk volume losses from dissolution are similar to those estimated from deformed shales in the Shimanto Belt in Japan (13 54% Kawabata et al., 2007), stylolites in Cretaceous to Palaeocene chalks in the North Sea (34 58% Safaricz & Davison, 2005), pressure solution in



Figure 8. Analysis from SEM EDS to quantify mass loss and resultant shortening strain within stylolite seams. (a c) Isocon plots show wt% calculated from bulk protolith and (a) Seam A, (b) Seam B, and (c) bulk stylolite, all areas outlined in (d). Numbers adjacent to elements are their multiplication factors. Dotted lines on isocon plots correspond to zero bulk mass change. Solid lines plotted through Al show actual bulk mass change, shown in percent in lower right of each plot. Horizontal error bars are 1 standard deviation of bulk protolith above and below stylolite; vertical error bars are 2σ from wt% analysis from EDS data. For details of mass change and strain calculations, see text. Plot (e) shows stylolite-hosted strain per recovered meter and stylolite frequency per recovered meter versus average CaCO₃ content within each core section. CaCO₃ content is from coulometric analysis performed during IODP Expedition 375. Filled points and lines correspond to the left axis and are colored according to depth downhole (see colorbar), colored points are strains per recovered meter calculated assuming a strain on each stylolite seam of 0.7 (e.g., a and b), whereas colored lines are strain ranges calculated assuming strains on stylolite seams from 0.1 to 0.9. Un lled black circles are stylolite counts per recovered meter. As we are dealing solely with shortening strains, we show them as positive.

a young marly thrust fault in the Spanish Pyrenees (14 44% Lacroix et al., 2015)• and stylolites in carbonates in the Permian Zechstein Basin in Germany (25 39% Koehn et al., 2016).

6.2. Strain in Stylolites

We now convert volume loss to shortening strain per individual stylolite and scale up our observations, using stylolite distribution, to determine the vertical shortening throughout Unit IV. As we are dealing exclusively with shortening strain, we express longitudinal strain values as positive for decreases in length throughout the remainder of this work to more clearly express the degree of shortening. Observational and chemical analyses of stylolites indicate uniaxial, vertical shortening by pressure solution, but quanti cation of this



strain is limited in scope to what is measurable within recovered core and by detailed study of selected, representative seams. We provide a limit for strain hosted by a singular stylolite seam by assuming all chemical volume loss is parallel to vertical stress (σ_{ν}). We then use an idealized stylolite, composed of seven 50 μ m wide seams of shortening 0.7 (a simpli ed approximation of the stylolite in Figure 8d), to calculate bulk uniaxial strain for recovered core within Unit IV. We present our data with error bars illustrating a range in calculated strain for each individual stylolite seam from 0.1 to 0.9 (Figure 8e). Bulk strain was calculated as change in length per recovered core length for each core section from holes U1520C and D using the equation:

$$\varepsilon_{\text{recovered}} = 1 - (L_{\text{recovered}}/L_o)$$
 (3)

where

$$L_o = (L_{\text{recovered}} - L_{\text{styl}}) + \left(\frac{L_{\text{styl}}}{1 - \varepsilon_{\text{styl}}}\right)$$
(4)

and $\varepsilon_{\text{recovered}}$ is the shortening strain in recovered core, $L_{\text{recovered}}$ is length of recovered core, L_o is the undeformed thickness of recovered core, L_{styl} is the thickness of stylolites within recovered core, calculated using the aforementioned idealized stylolite and the frequency of stylolites observed within recovered core, and $\varepsilon_{\text{styl}}$ is the shortening strain on an individual stylolite (0.7 from isocon plots and porosity loss; Figures 8a and 8b). Where core sections spanned the contact between two lithostratigraphic units, the percentage recovery was assumed to be the same either side of the boundary. The resultant shortening for each core section is plotted per recovered meter of core against average CaCO₃ content for the core section in Figure 8e. Also plotted is stylolite number per recovered meter within each core section, which shows an approximately exponential increase in strain as CaCO₃ contents increase above 70%. Increased stylolite-hosted strain with increasing CaCO₃ content is also particularly clear in the deeper sections of the core (Figure 8).

7. Controls on Fault and Stylolite Distribution

Both faults and stylolites cluster when spaced <1 m apart (Figure 7), meaning they are more likely to form near an existing feature than elsewhere (Gillespie et al., 1999). Clustering of features therefore suggests either: (1) feature formation changes conditions to make further nearby feature formation more likely, or (2) conditions in areas hosting many clustered features must be preferential for their formation. While some factors controlling the distribution of faults and stylolites are shared (differential stress, fluid pressure, etc), the distinct nature of faulting and pressure solution (i.e., physical vs. physiochemical) means that many of the factors controlling fault and stylolite distributions are distinct. Faults likely span several small-scale lithological boundaries with far greater lengths than stylolites and may have initiated in different lithologies before propagating through the core. We therefore refrain from suggesting any bulk controls on faulting based upon observations from drill core and focus on stylolites and stylolite-fault interaction.

7.1. Controls on Stylolite Formation

Because numerous factors influence stylolite formation (e.g., Toussaint et al., 2018), despite the large range of properties measured by IODP Expeditions 372 and 375 (Wallace et al., 2019), a peak in stylolite frequency does not always coincide with a measured change in core properties. We therefore do not attempt to fully characterize all factors which could influence stylolite formation but rather suggest some indicative variations which correlate with stylolite frequency.

7.1.1. Mineralogical Control on Dissolution

At a macroscopic scale, stylolites in Unit IV form anastomosing networks with low-amplitude, long wavelength roughness (Figure 3a). This is probably due to the lack of distinct mineralogical heterogeneities in a sediment dominated by carbonate and clay (Ebner et al., 2010; Railsback, 1993). Sediments in Unit IV rarely host more than 20% quartz and feldspar, and there are no discernible peaks in stylolite frequency correlating solely with an increase in quartz, feldspar, or clay content (Figure 6). In contrast, stylolite frequency within clusters shows a correlation with CaCO₃ content throughout Unit IV (Figures 6 and 8e). This is logical as calcite, which the CaCO₃ probably represents, is more soluble than the other minerals present in Unit



IV at the in situ P-T conditions (Heald, 1955; Plummer & Busenberg, 1982) and appears to be the main dissolved material (Figures 3d and 3e). Having said this, stylolites are not ubiquitous throughout all high $CaCO_3$ content lithologies, suggesting that high $CaCO_3$ content influences the intensity of dissolution rather than the location of stylolite formation. This $CaCO_3$ content control on dissolution magnitude likely exists between a lower bound where its presence allows dissolution and an upper bound where there is not enough phyllosilicates to form a stylolitic residue and localize pressure solution (cf. Zubtsov et al., 2004). This is reinforced by the total absence of stylolites in the mudstone, where $CaCO_3$ content is very low, the upper bound, as observed after dissolution is likely <95 wt% based upon observations of abundant stylolites within the $CaCO_3$ -dominated chalk at the base of Unit IV.

7.1.2. The Relationship of Stylolites and Lithology

Stylolites are most common in the deepest chalk unit, ~650 out of 873 total stylolites occur there, and as CaCO₃ content is high throughout the chalks, this stylolite clustering requires an additional factor to explain heterogeneous stylolite development in space. On a submillimeter scale, stylolites dissolve relatively large $(<200 \,\mu\text{m})$ fossils throughout the sediments on discrete localized seams (Figures 3d and 3e). Fine clay leftover from dissolution is preserved within some stylolites as ne, elongate, grains with seam-parallel long axes (Figure 3e). Dissolution of a grain would continue until the grain contact area becomes so large that stress on that contact is below a critical value that is needed for pressure solution to operate, and grain boundary healing occurs (Rutter, 1983; van den Ende et al., 2019). Over the stress history of the sediments, grain contact surfaces probably dip below this critical value when the grain is elongated parallel to the seam, developing a shape-preferred orientation (Figures 3d and 3e). Such a shape-preferred orientation adjacent to stylolites has also been reported by many previous workers (Durney, 1972; Ebner et al., 2010; Toussaint et al., 2018). This interaction could explain the undulating nature of stylolites around and between fossils as they follow the path of maximum positive normal stress perturbation relative to the background (Figures 3d and 3e). Coupled with the accelerating effect of clay-rich and mineralogically diverse lithologies on dissolution (Aharonov & Katsman, 2009; Hickman & Evans, 1995), a clay-rich horizon with fossils interacting to concentrate stress would produce more rapid development of stylolites than a lithology without either clay or fossils. Indeed, Railsback (1993) shows that abundant contrast of clasts with a clay-rich matrix would host greater dissolution than either more abundant clasts or a more clay-poor matrix.

In contrast to fossiliferous lithologies, burrow margins within bioturbated lithologies host more well-developed stylolites and are more enriched in insoluble clay than both the internal volume of the burrow and the surrounding material (Figure 3c). This enrichment suggests that where an initial concentration of clays or other insoluble material exists, dissolution is more localized and forms zones of well-developed stylolites, in agreement with observations from experimentally deformed diatomite and gypsum-illite composite by Gratier et al. (2015) and modeling by Aharonov and Katsman (2009) and Hickman and Evans (1995). Bioturbated and fossiliferous lithologies therefore have contrasting effects on stylolite development; bioturbated lithologies concentrate fewer, more intense, stylolites locally on burrow margins, while fossiliferous lithologies host many, less intense, more spaced stylolites.

7.1.3. Porosity Around Stylolites

Porosity is lower in stylolites than in the immediately adjacent material (Figure 4), suggesting porosity is lost by material dissolution causing closer grain packing (Macente et al., 2018; Pluymakers & Spiers, 2015). Larger porosity losses in more porous sediments (19R1W in Figure 4) show stylolites develop better in ner grained, higher porosity sediments. Similar to previous workers, we nd zones of increased porosity immediately adjacent to stylolites (Figure 4, Heap et al., 2018). This is consistent with smaller grain contact areas supporting higher stresses, therefore localizing dissolution. Microfossils in stylolites are partially dissolved, showing porosity reduction where stylolites developed within sediments hosting fossil-bound void space (Figures 3 and 4). Absence of overgrowths adjacent to stylolites indicates that calcite did not precipitate immediately adjacent to the natural stylolites studied here, suggesting either precipitation rates were prohibitively low (i.e., precipitation-limited pressure solution), or there was limited solute available for precipitation. Diffusion-limited and precipitation-limited pressure solutions have been reported at low temperature (<150°C) for uniaxial compression of calcite at low (<0.04) and moderate strains (>0.04), respectively (Zhang et al., 2010). Although not well constrained, we raise the possibility of higher stylolite-marginal porosity promoting throughflow of fluids, allowing faster solute transport away from sites of dissolution and effectively limiting precipitation immediately adjacent to stylolites. Throughflow of fluids would likely



freshen pore water at depth by lateral flow. This is not reflected in downhole chlorinity data at the speci c depths where we observe stylolite clusters, but pore water sampling was too sparse to identify localized flow around individual stylolite clusters (Wallace et al., 2019). In concert with reduced grain size and slight shape-preferred orientation bordering stylolites (Figure 3), dissolution may also have occurred within a volume surrounding the stylolite, as previously suggested by Ebner et al. (2010). The outward propagation of the dissolution front from a clay-rich seam coincident with porosity reduction within stylolites has also been recognized by Macente et al. (2018) in experiments on a halite-biotite composite. This may explain how more well-developed stylolites grow to encompass several seams of phyllosilicates (Figure 3).

7.1.4. Temporal and Spatial Variations in a Stylolite Population

Given the lithological, physical, and chemical controls on the self-organising process of stylolite formation, the nal distribution in Unit IV reflects spatial changes in carbonate and clay content, porosity and sedimentary structure, and grain-scale stress concentration and relative solubility (Figure 6), in decreasing order of importance. It seems the stylolite population within Unit IV is controlled dominantly by protolith properties rather than by stylolite formation altering initial conditions. The reasons for this are that stylolite seams (1) individually appear to have very localized regions of influence, only altering grain shape within tens of micrometers (Figures 3d and 3e), (2) form very closely without directly interacting, as would be expected if exerting a strong control on stylolite formation (Figures 3b and 3c), (3) correlate in frequency with peaks in bulk properties such as $CaCO_3$ content (Figure 6), and (4) occur in frequency peaks which are randomly distributed over the length of Unit IV when spaced >1 m apart (Figure 7). It has been postulated that the concentration of clays during stylolite formation accelerates and localized further dissolution, meaning stylolite formation may cause further stylolite formation there (i.e., strain localization) (Aharonov & Katsman, 2009; Gratier et al., 2015; Hickman & Evans, 1995; Macente et al., 2018). This self-organization means that minor initial perturbations in the conditions outlined earlier are ampli ed once dissolution on stylolites begins. The subsequent temporal evolution of a stylolite is therefore conditional upon every previous feedback of evolving conditions with dissolution on that stylolite. It is entirely possible that when stylolites occur within tens of micrometers of one another, they are also influenced by changes in conditions resulting from adjacent stylolites, attested to by increasing clustering of closely spaced stylolites (Figure 7), but this is not a control on the macroscale distribution of stylolites.

Comparing stylolite frequency from 509 to 848 mbsf at Site U1520 to that from 790 to 1,350 mbsf on the Ontong Java Plateau to the north (Lind, 1993), it is clear a far higher frequency of less well-developed stylolites is present at shallower depths at Site U1520. Lind (1993) describes an erratic, highly localized, stylolite frequency, overall increasing in abundance with depth. Stylolites are said to be irregular in size and localize along burrows and clasts of contrasting mineralogy. This description is similar to what is seen at Site U1520, where randomly distributed peaks in stylolite frequency increase in intensity with depth (Figure 6), burrows localize intense stylolite-rich regions along their margins (Figure 3c), and more fossiliferous regions host more spaced, less well-developed stylolites (Figure 3c). Carbonate content in the Ontong Java Plateau sediments is >90%, similar to the chalk at the base of Unit IV (Wallace et al., 2019). The difference in frequency is probably a result of a contrast in the intensity of stylolite development at each site; the limited depth range of Unit IV (509 848 mbsf) means that stylolites at Site U1520 are less well-developed than they would be at the depths studied by Lind (1993). Our counting is therefore more likely to be sensitive to less well developed stylolites, leading to a higher frequency. Regardless, controls on stylolite formation appear to be similar at both sites, suggesting that our observations of calcareous-pelagic sequences are valid beyond the Hikurangi sedimentary input sequence.

7.2. Interactions Between Stylolites and Faults

Microscale observations of faulting within Unit IV were limited, but faults appear to coincide with clay-rich weak layers prior to later alteration of their damage zone (Figure 5). Thick (3 mm) clay-rich seams, concentrated by pressure solution, are present astride normal faults. These seams likely formed prior to fault formation, based on good preservation of the fault surface and the low strain rates at which pressure solution occurs relative to brittle fault formation by fracturing (Figure 5, Gratier et al., 2013; Paterson & Wong, 2005). The brittle faulting therefore probably localized within pre-existing clay-rich seams, exploiting thin zones of relative weakness from previous viscous deformation (Viti et al., 2014). Both fault textural and distribution data, therefore, suggest stylolites are at least partially responsible for fault localization in Unit IV.

Where brittle fault surfaces are localized on strain-weakened low-angle surfaces, slip would result in centimeter-scale damage zones comprising both tensile and shear fractures adjacent to the slip surfaces (Figure 5). These fractures represent new fluid pathways, rapidly increasing the surface area on which pressure solution can occur (Gratier et al., 1999). Strain rates on individual stylolite seams studied here (of strain 0.7; Figures 8a and 8b) range from 3.7×10^{-16} to 2.2×10^{-14} s⁻¹ for ages of 60 and 1 Ma (Figure 2d), respectively. Although these are at the slow end of typical average geological strain rates in deforming zones (Fagereng & Biggs, 2019; P ffner & Ramsay, 1982). Our calculations of strain from stylolite frequency likely underestimate strain as it does not account for dissolution outside stylolite seams.

8. Modeling Uniaxial Strain Accommodation by Pressure Solution

The age, depth, and chemical constraints on the stylolite population described in this study provide a unique opportunity to compare strain accommodation from published constitutive models of pressure solution (Rutter, 1976; Spiers et al., 1990, 2004; Zhang et al., 2010) to strains calculated from direct observations of core from Site U1520 (section 6.2). The population of stylolites described here has been well characterized under well-de ned P-T conditions, providing a directly comparable natural analog for the pressure solution described in the aforementioned models. We therefore use data from Site U1520 to construct a history of physical conditions and strain from intergranular pressure solution for the ~60 Myr history of Unit IV.

As pressure solution is rate limited by the slowest of the serial processes of dissolution, diffusion, or precipitation, it is important to select the most appropriate model. The presence of partially dissolved fossils suggests dissolution is not the rate-limiting factor for pressure solution on stylolites (Figure 3), but as transitional diffusion or precipitation-limited behavior has been reported for calcite under conditions analogous to those studied here (Zhang et al., 2010), discerning the dominant rate-limiting factor from diffusion and precipitation is less clear. Based upon the observed lack of overgrowths adjacent to stylolites, we earlier inferred fluid flow and solute removal away from sites of dissolution in a system that is at least partially open (section 7.1.3), limiting local reprecipitation. If solute was rapidly removed from sites of dissolution, geochemical gradients would have been high at sites of dissolution and diffusion-limited pressure solution would have been faster than modeled in a closed system. We therefore use equations for diffusion-limited pressure solution as a lower bound on strain rate. As the number of assumptions within the modeling is large, we limit our discussion of absolute strain magnitudes, instead focusing on the location of pressure solution localization onto stylolites and which factors within the modeling could be better constrained.

We use the age model described earlier and reported in Wallace et al. (2019) to calculate past sediment depth at each depth and time step before calculating temperature and vertical stress using density data and a modern geotherm derived from measurements collected at the site (Figure 2). We calculate stress conditions for pore fluid factors of 0.4 and 0.95 for comparison. Solubility (Plummer & Busenberg, 1982) and diffusivity (Nakashima, 1995) are calculated from estimated temperatures (20 33°C; section 3.2) and are used with a relatively thin grain boundary fluid lm thickness of 1×10^{-9} m (Renard et al., 1997) and an equation accounting for measured porosity at each depth (Pluymakers & Spiers, 2015) to calculate strain rate using published equations for diffusion-limited pressure solution (Rutter, 1976; Spiers et al., 1990, 2004; Zhang et al., 2010). We scale this modeled strain rate by the calcite content at that depth, both linearly and following the scaling relation of Zubtsov et al. (2004) (see section 8.2). Full details of the modeling methodology are available in Texts S1 and S2 in the supporting information.

Several assumptions are made within our history of physical conditions. Not least that data collected at a point during IODP Expedition 375 are representative for density, composition, porosity, and temperature gradient in the incoming sequence. We extrapolate these values back in time using an age model based upon macrofossils, microfossils, and palaeomagnetic reversal ages, all of which have varying degrees of dating error (Figure 2d). The only assumed constant condition throughout the history of the sediment is the modern temperature gradient from the seafloor downward, applied at each time step during the history of deposition. The main consequence of the temperature gradient is variability in the solubility of calcite, which is inversely correlated with temperature (Plummer & Busenberg, 1982). A higher temperature gradient, possibly due to higher heat flow from younger underlying oceanic crust (Villinger et al., 2002), would therefore reduce calcite solubility early in the sediment history. Therefore, our values probably overestimate



calcite solubility in the early history of the sediment. Further to this, no attempt is made to quantify calcite solubility with increasing stress due to the lack of empirical published data.

Following constituitive equations for diffusion-limited pressure solution, grain-scale diffusivity effects within the model are captured within the diffusion coef cient and mass transfer width terms (Rutter, 1976, 1983). A simple Arrhenius temperature dependency (Nakashima, 1995) is unlikely to capture the complexity of electrochemical and grain boundary effects on the diffusion coef cient (D), while a single value for the mass transfer path width (S) ignores the complexity of subgrain size fluid pathways throughout grain boundaries between like and unlike minerals (Renard et al., 1997; Zubtsov et al., 2004). Furthermore, these values are likely to evolve with time and strain as well as the physical background conditions modeled here, possibly explaining the difference between our modeled and observed strains (Figure 11). We note that we have used a low-end value for the grain boundary fluid Im thickness and current rather than initial porosities these assumptions will tend to underestimate strain rates. Porosity within the sediments is present within and without spherical fossils (Figure 4); meaning, porosity estimates from the core are likely greater than what is realistically available for precipitation of solute, if such precipitation does not occur within shells. The intrafossil porosity is not accounted for in the model, and the resulting strains are therefore likely to be overestimated. We consider this to be of relatively minor importance given the number of assumptions in the modeling and choose to focus our discussion on the distribution, rather than magnitude, of strain from pressure solution.

8.1. Modeling Results and Pressure Solution Controls

Modeled strain rates and cumulative strains are shown for the top and bottom of Unit IV in Figures 9e and 9f and for the whole of Unit IV in Figures 10c and 10d. Throughout the history of the sediment the dominant control on the modelled strain rate is the overlying sediment depth (Figure 10a). Sediment depth is controlled by the age model for the site (Figure 2d) and expressed in the modelling by effective normal stress (Figures 9d and 10a) and temperature (Figure 9b). After initial deposition, slow accumulation of sediment caused strain rate to increase gradually in the deepest 125 m of Unit IV during the rst 50 Myr of the modeled history (from >10^{-16.5} to ~10⁻¹⁶ s⁻¹; Figures 2e, 9e, and 10c). These strain rates are relatively slow by geological standards (Fagereng & Biggs, 2019; P ffner & Ramsay, 1982) but were sustained long enough (until 15 Myr ago) to accommodate strains of up to ~0.08 before the upper ~215 m of Unit IV were deposited (Figures 9a, 9f, and 10e). Around 15 Myr ago, the age model dictates continuous sedimentation rate increases up to the present (Figure 9a). This manifests as fluctuating, relatively rapid increases in modeled strain rate (up to ~10^{-15.5} s⁻¹) over ~14 Myr, until rapid loading of turbidites in Units I III in the last ~1 Myr increases strain rates further (up to ~10⁻¹⁵ s⁻¹; Figures 9e, 9f, 10c, and 10d). This strain rate increase has the most marked effect in the deepest 100 m of Unit IV, where sediments are over 25 Myr old (Figures 9f and 10d).

At a given time-step modeled strain rates generally increase with depth, fluctuating locally because of local variations in porosity and calcite content (Figure 10d). Strain rates are highest between 825 and 850 mbsf, where calcite content is high, with local maxima coinciding with large fluctuations in measured porosity (e.g., \sim 830 and \sim 840 mbsf in Figure 10c). Unsurprisingly, this depth range of higher modeled strain rates, and the local maxima within it, has the highest modeled strain within Unit IV (Figure 10d). Final strains at the local maxima are up to 0.05 above the background within this depth range, even when the lower strain sediments are >10 Myr older (\sim 840 vs. \sim 845 mbsf in Figure 10d).

The constant geothermal gradient applied during the modeling means temperature-dependent factors effectively become depth dependent, and the variation of solubility and diffusivity with temperature can be discussed alongside that of depth-dependent factors such as effective normal stress. During the rst 50 Myr of our model, the deepest calcareous sediments (>700 mbsf) maintain low effective normal stresses (<2 MPa) and temperatures (<10°C), corresponding to high solubilities (>10^{-5.6} m³ m⁻³; Figures 9b 9d, 10a, and 10b). The strain accumulated over this period (up to 0.08) is not negligible compared to nal totals in our model (up to 0.15), but rapid loading in the more recent history of the model increases strain rates from ~10⁻¹⁶ s⁻¹ to around 10⁻¹⁵ s⁻¹, adding an extra strain of 0.07 over the nal 15 Myr of Unit IV at average strain rates of approximately $10^{-15.6}$ s⁻¹. This is similar to the accumulated strain over the previous ~50 Myr and, while not surprising, illustrates how a nonlinear history of sedimentation or stress loading may affect rates of pressure solution during the history of deformation. Changes in effective normal stress, reflected in the modeled strain rate (Figure 10c), are even more apparent when pore fluid pressure is





Figure 9. Pressure solution model inputs and outputs with time. Plots show (a) sediment depth, (b) temperature and solubility calculated from Plummer and Busenberg (1982), (c) lithostatic and hydrostatic stress, (d) effective normal stress, (e) strain rate from diffusion controlled dissolution of calcite according to published models for pressure solution (Rutter, 1976; 1983), and (f) cumulative strain from diffusion controlled dissolution of calcite at the top and bottom of Unit IV with time. Inset in (f) is zoomed area showing cumulative strain at top of Unit IV. Most plots show results for the bottom (848.45 mbsf) and top (510.96 mbsf) of Unit IV. Plots of are for a grain size of $100 \,\mu$ m. As we are dealing solely with shortening strains, we show them as positive.

increased to near-lithostatic levels, reducing the stress driving dissolution and the modeled strain rate, and limiting the resultant strain (Figure S1).

8.2. Comparing Modeled and Calculated Strains from Pressure Solution in Unit IV

Our calculation of stylolite-hosted strain within Unit IV is based on stylolite frequency and strain on each individual stylolite seam (Figure 8), meaning our total calculated strain reflects stylolite frequency in core image logs (Figures 6 and 10). This stylolite-hosted strain is likely a minimum value for strain within the whole of Unit IV, as pressure solution is likely to act upon much of the depositional layering as a diagenetic process (Gratier et al., 2015), and we have not quanti ed strains caused by granular flow or faulting. Modeled strains, however, are calculated using equations derived from experiments of distributed dissolution and bulk shortening rather than that localized on stylolites (Rutter, 1976, 1983; Zhang et al., 2002, 2010). Strains are therefore comparable within the depth range of stylolite clusters or over length scales encompassing several clusters (>1 m). Modeled strain from pressure solution at hydrostatic fluid pressure is almost always higher than observed strain, each with peaks which rarely coincide (Figure 11a). When studied at the length scales of lithological units (10s to 100s m) or core sections (\sim 10 m), calculated and modeled strains rarely agree, with modeled strains almost always exceeding the measured magnitude of strain (Figures 11b and 11c). At near-lithostatic pore fluid pressures, the model reproduces the average strain magnitude when compared over larger length scales (Figures 11e and 11f) but does not reproduce observed stylolite clustering, giving lower than measured strain within stylolite clusters and higher than measured strain between clusters (Figure 11d). As modeled strain uses bulk properties throughout the core and stylolites host highly localized dissolution, it seems that stylolite formation is controlled by local perturbations not accounted for in our models. This is not surprising as we cannot model unknown initial heterogeneities





Figure 10. Calculated conditions for the full depth of Site U1520 with time. Colored plots show (a) effective normal stress, (b) calcite solubility, (c) strain rate, and (d) cumulative strain with depth and time. Gray lines on color plots show contacts between simpli ed lithological units within Unit IV; model results are for a grain size of 100 μ m. As we are dealing solely with shortening strains, we show them as positive.

in the sediment column, and our models are therefore averages over multiple clusters. Despite our attempt to integrate the evolution of some of the parameters controlling stylolite evolution, the use of the constitutive equations of pressure solution (Gratier et al., 2013; Rutter, 1976, 1983; Spiers et al., 1990, 2004) conceived and calibrated with constant or controlled parameter values does not accurately predict the location or magnitude of cumulative strain of intensive dissolution on stylolites. Beside the possible uncertainty on the values of the various parameters of the equations, the evolution of the parameter values with stylolite evolution is not yet enough constrained within our model, particularly the effects of self-organized processes.

Porosity is variable both within Unit IV and locally around stylolites (section 7.1.3 and Figure 4). The coincidence of observed stylolite clusters and localized porosity peaks (e.g., at ~760 and ~805 mbsf) is consistent with observations of greater volume loss and increased resultant shortening strain in higher porosity zones (19R1W in Figure 4). Modeled strains have local high-frequency variations with depth due to variable porosity and calcite content measured throughout Unit IV (Figures 6c and 6d). With depth, some peaks in cumulative modeled strain coincide with peaks in stylolite-hosted strain (Figures 11a and 11d). Previous studies that show strain increases with increased contact stress over reduced grain contact area (Niemeijer et al., 2002; Pluymakers & Spiers, 2015; van-den-Ende et al., 2018; Zhang et al., 2010), suggesting that applied variations in strain rate with porosity may be, at least partially, representative of variations in pressure solution within natural basins containing calcareous sediment.

Composition is another important difference between the sediments studied here and the materials used to study the dissolution of pure calcite (Rutter, 1976, 1983; Zhang et al., 2002, 2010). Our attempt to account for the heterogeneous mineralogy of Unit IV by limiting strain from dissolution to soluble grains does not account for (1) the accelerating effects of electrochemical potential on pressure solution across mineralogically heterogeneous grain boundaries or (2) stylolite localization associated with increased clay fraction at particular horizons (Aharonov & Katsman, 2009; Greene et al., 2009; Hickman & Evans, 1995). Furthermore, stylolites are typical of pressure solution in dominantly calcareous sediments, whereas





Figure 11. Comparison of modeled and calculated strains for compaction by stylolites at hydrostatic or near lithostatic fluid pressures within Unit IV. Plots show (a and d) nal strain with depth per meter, (b and e) strain within each lithology and all of Unit IV (outlined in black), and strain within each core section (c and f) for hydrostatic (a c) and near-lithostatic (d f) pore fluid pressures. Error bars show strain ranges for individual stylolite seams of strain 0.1 0.9 used in calculation of bulk strain. As we are dealing solely with shortening strains, we show them as positive.

pressure solution within more clay-rich sediments would likely occur on depositional layering, accommodating vertical shortening without forming stylolites (Gratier et al., 2015). Evidence for this pressure solution in more clay-rich lithologies would be dif cult to observe and has not been counted here as a straining feature; meaning, our measurements of stylolite-hosted strain are likely a minimum value for the whole of Unit IV. Previous experiments on pressure solution in two-phase mixtures found a linear scaling relationship exists between strain rate and soluble species content up to 50%, where it plateaus until 70% soluble species, before decreasing linearly at soluble species contents of 70 100% (Zubtsov et al., 2004). This was thought to be because diffusion in a fluid phase trapped along grain contacts between different minerals is faster than along grain contacts between the same mineral. The mineral assemblage varies signi cantly throughout Unit IV (Figure 6) and variations in grain-scale factors such as these could account for variations in observed strain that have not been recreated in our model. We apply the scaling relation of Zubtsov et al. (2004) but nd it does not better recreate calculated stylolite-hosted strain distribution or magnitude (Figure S2). Better quantitative constraints to measure and model the effect of grain-scale processes on macroscopic strain, by both distributed and localized pressure solution, would aid the ability of models to recreate observed trends in strain.



Cumulative strains modeled over the history of Unit IV vary signi cantly in magnitude from those estimated from stylolite mass loss and frequency. Model results for grain sizes of 10 and 100 μ m vary over three to four orders of magnitude, and agreement of calculated and modeled strains is only achieved at hydrostatic fluid pressures when the modeled grain size is 100 µm (Figures 11b and 11c) and at near-lithostatic fluid pressures when the modeled grain size is $50 \,\mu\text{m}$ (Figures 11e and 11f). The mean grain size measured in the chalks in Unit IV is $<40 \,\mu$ m, with larger microfossils 200 μ m (Figure 3), suggesting that either (1) the model overestimates strain rate from grain dissolution or (2) near-lithostatic pore fluid pressures have been sustained for long periods of the sediment history and (3) not all pressure solution forms stylolites or (4) strain within Unit IV is controlled by the dissolution of grains larger than those comprising the matrix (Figures 10 and 11). The dissolution of microfossils seen in microstructural observations suggests that the latter may be the case, though matrix grains also appear to be dissolved, showing a shape-preferred orientation adjacent to stylolites (Figure 3). Alternatively, the ne-grained nature of the sediment, and possible fluid compartmentalization by overlying low permeability mudstones (Figure 1b), may locally lead to greater than hydrostatic pore fluid pressures during the history of the sediment, reducing effective vertical stress and subsequent dissolution. However, there is no other evidence, such as fractures, veins, or fluid escape structures, indicative of sustained near-lithostatic fluid overpressures at Site U1520 (Wallace et al., 2019). A further point for consideration is that the diffusive distance (d) is assumed to be equal to the grain size in our model. It may be that in systems where lateral long-range solute transport may be operating (e.g., in an open system with advective transport of calcite) diffusive distance can be larger than grain size (Dewers & Ortoleva, 1990; Gundersen et al., 2002; Lehner, 1995).

8.3. Forward Modeling Pressure Solution to the Plate Interface

Having modeled pressure solution in Unit IV through the last ~ 60 Myr to the present day, we also extend the dissolution model to show the nal effect of pressure solution when the sediments reach the toe of the Hikurangi plate interface. We use the thickness of sediment overlying Unit IV from Site U1520 to the toe of the thrust on seismic line 05CM-04 to calculate vertical stress into the future (Figure 1 Barnes et al., 2020). Strain rate from pressure solution is calculated as previously.

The thickness of sediment added is less than 500 m at the toe (Figure 1), reflected by a change in normal stress of ~4 MPa (Figure S3). Similarly, a minor change in calcite solubility results from higher temperature with increased depth. The distance between Site U1520 and the toe of the megathrust is relatively short (~20 km), occurring rapidly compared to the previous model and hosting shortening strain increases of less than 0.025 throughout Unit IV (Figure S3). The small difference in strains indicates that the majority of pre-subduction dissolution-hosted strain has already been accommodated within Unit IV over the last (~60 Myr) so that stylolites observed at Site U1520 are likely very similar to those existing within Unit IV at the toe of the megathrust.

9. Future Subduction of Unit IV and Rheology of the Current Plate Interface

Modeled and observed strains from pressure solution in Unit IV at site U1520 are comparable when averaged over tens of meters and using a diffusion length scale comparable to the coarsest sediment fraction. Forward modeling to the toe of the megathrust shows only a slight change in strain, suggesting the majority of pre-subduction compaction had already been accommodated by the time the sediment reached Site U1520 (Figure S3). Depth-converted seismic pro les indicate the lower part of Unit IV is likely the protolith for the localized plate interface, where it is likely to accommodate shear strain during subduction (Barnes et al., 2020).

9.1. Rheological Variation Throughout Unit IV

Unit IV is highly variable with regard to mineralogy, porosity, and lithological texture (Figure 6). These factors, along with partial lithi cation and abundant pore water, are likely to influence slip and deformation style and contribute to a heterogeneous plate interface (Barnes et al., 2020; Delle Piane et al., 2017; Mittempergher et al., 2018; Sample, 1990; Skarbek & Rempel, 2017; Wallace et al., 2019). Compounding this, the calcareous-pelagics in Unit IV overlie a sequence of volcanic conglomerates, completely rheologically distinct and even more locally variable (Barnes et al., 2020; Wallace et al., 2019). The rheology of intact lithologies comprising Unit IV at the plate interface would therefore already be locally variable, but pre-subduction strain on faults and stylolites will cause local changes in rheology from those predicted for the intact sediments of Unit IV.

Strains calculated from stylolites within Unit IV are not large. Uniaxial shortening strain for the entirety of Unit IV is ~0.004 but represents local accumulation of insoluble clays within each discrete <1-mm-thick stylolite (Figure 3). Each stylolite does not host enough insoluble clay to signi cantly alter the rheology of the bulk rock, but where stylolites cluster the accumulation of clays within stylolites may become locally significant (Figure 6). Stylolites cluster when less than 1 m apart (Figure 7), with clusters hosting up to 80 stylolites per meter (Figure 6). Clay concentrated on stylolites in these clusters represents strain-weakened horizons from early viscous deformation due to the weakness of clay relative to the encompassing carbonate (Boulton et al., 2019; Rabinowitz et al., 2018), possibly signi cant enough to locally alter bulk deformation style. Indeed, some faults in Unit IV are localized along clay-rich seams of similar thickness to those within stylolites (Figures 3 and 5). Clusters of stylolites increase in intensity with depth, increasingly altering lithological texture and resultant deformation style on increasingly interconnected anastomosing stylolite seams (Figures , 3, 6, and 12).

Faults within Unit IV are rheologically variable, with weak surfaces between two moderately weak zones of damaged material (Figures 5 and 12). Individual fault and fracture zones are more commonly considered than stylolites as zones of strain weakening, likely to host later deformation (Ferraro et al., 2018; Leah et al., 2018; Mitchell & Faulkner, 2009), but clustering of faults in Unit IV makes contemporaneous reactivation of several faults within a centimeter- to meter-thick volume likely at the plate interface (Figures 6, 7, and 12). In addition, pressure solution has concentrated phyllosilicates on fractures in damage zones, concentrating clays and further weakening areas hosting fault clusters (Figures 5 and 12).

9.2. Future Décollement Location and Deformation Localization

Within Unit IV, clusters of clay-rich stylolite seams or fracture-bound faults represent these low-volume fraction phases with lower shear strengths than the host rock, likely to localize deformation at the plate interface. Rheological heterogeneity is another key factor in deformation localization during reactivation of pre-existing weaknesses (Imber et al., 2008; Leah et al., 2018; Lyakhovsky et al., 1997; Willemse et al., 1997), and stylolite and fault clusters represent heterogeneous intervals throughout Unit IV. One particular interval, near the base of Unit IV (835 845 mbsf; Figure 6), hosts nearly 400 stylolites in two large clusters, increasing the volume fraction of clay present by localized pressure solution (Figure 6). This increased clay fraction creates a shear strength contrast within the sediments, making strain more likely to localize there when subject to shear stress. Shallower in the sediment stack, mudstones within Unit IV (e.g., ~720 740 mbsf) may also be signi cantly weaker layers likely to localize shear strain. The localization of deformation likely to occur upon reaching the plate interface would manifest as incrementally greater shear strain is accommodated in the weaker horizons rather than in the stronger intervals, creating a thin cumulative volume hosting high shear strain, including a likely horizon near the base of Unit IV (Figure 12).

9.3. Deformation Mechanisms at the Plate Interface: Implications for Slow Slip

Slip on faults and fractures within their damage zones will be frictional, with properties controlled by that of the material on the fault. Where faults have localized upon pre-existing stylolites, therefore, slip character is likely to be controlled by phyllosilicates localized in the stylolite seam (Figure 5). Where faults have formed away from stylolites, frictional slip is likely to reflect the bulk protolith, dominantly calcite (Figure 6). Calcite and clays are likely to mix during slip, possibly by weak phase smearing (Rutter et al., 2013). Friction on these surfaces would likely be dominated by phyllosilicate-hosted slip with a lesser component of calcite. This would lead to a mixture of velocity strengthening and weakening conditions over varying conditions and could provide appropriate conditions for unstable slow slip on the plate interface (Boulton et al., 2019; Rabinowitz et al., 2018; Tesei et al., 2014).

Localized slip surfaces are also likely to form on weak, clay-rich stylolites, where they could accommodate shear strain by a combined frictional-viscous flow mechanism (Bos & Spiers, 2002; Gratier et al., 2013; Willemse et al., 1997). This entails frictional slip on an anastomosing phyllosilicate foliation with removal of material from grains within the foliation by pressure solution, similar to deformation modeled for quartz-bearing gouge by den Hartog and Spiers (2014) or less phyllosilicate-rich halite or quartz-bearing gouge by Bos and Spiers (2002). As calcite solubility is high at low temperatures (Plummer & Busenberg,





Figure 12. Schematic overview of strain localization and weakening in Unit IV at Site U1520 and its possible future rheological behavior during subduction. Sketches with notes (above) show locally important textures and rheological notes from Unit IV at various stages during subduction (middle). Thermal gradient (below) is from Antriasian et al. (2018); resultant calcite solubility is calculated using the method of Plummer and Busenberg (1982).

1982), abundant calcite within these deforming zones will likely deform by pressure solution at the shallow plate interface. Intensive pressure solution would cause volume loss from the volume accommodating shear between clasts, possibly increasing relative clay volume fraction within the sliding portion of the plate interface shear zone and altering the shape of clasts (Gratier et al., 2015), thereby causing strain weakening. Alongside this, carbonates experience signi cantly more variable deformation at relatively low temperatures than their siliciclastic counterparts, including recrystallization (Kennedy & White, 2001), crystal plasticity (Verberne et al., 2013), and variable frictional properties (Verberne et al., 2014).

Within a plate interface shear zone hosting grain size reduction of heterogeneous materials over varying depths, pore fluid pressures, and temperatures (Barnes et al., 2020), it is likely that pressure solution will be locally variable on the submillimeter scale, removing and redepositing material locally around grains or clasts. At shallow depths, the temperature gradient of the northern Hikurangi Margin is thought to be relatively high (Antriasian et al., 2018), corresponding to a rapid reduction in calcite solubility as materials are buried (Figure 12). The ability of pressure solution to cause removal of calcite, therefore, is greatly reduced outside regions of low temperature, likely corresponding to less than ~150°C or 10- to 15-km depth (Figure 12 Antriasian et al., 2018). This is a similar depth to the deeper end of the slow-slipping zone, possibly indicating that pressure solution of calcite may decrease in relative importance as a mechanism accommodating deformation at the plate interface outside the area recognized to host slow slip.

10. Conclusions

Unit IV at Site U1520 hosts clustered faults and stylolites within muddy pelagic carbonates with variable composition. The frequency of stylolites increases with depth, whereas faults are more common within several shallower horizons. Stylolite frequency increases approximately exponentially with CaCO₃ content.

Tectonics



Stylolites comprise anastomosing individual seams, each hosting mass losses of 11 45% within a horizon of lower porosity compared to the host rock. Uniaxial shortening, accounting for porosity change, is high on individual stylolite seams (~0.5) but small over greater distances because localized horizons host dissolution. Stylolites concentrate clays by dissolution, forming localized clay-rich surfaces within a mostly intact host rock. Faults in Unit IV are present in more lithi ed units, locally exploiting pre-existing stylolites. Slip surfaces host adjacent centimeter-scale damage zones which have themselves undergone pressure solution, showing that mixed brittle and viscous deformation occurs at very shallow depths in calcareous sediments. Stylolite distribution reflects an interplay of grain-scale physical and larger scale compositional factors, forming most commonly where calcite content is high, clay is present to accelerate dissolution, or high initial porosities localize dissolution.

We model pressure solution using published models for intergranular pressure solution of carbonates over the depositional history of Unit IV. To do this, we determine an absolute P-T-t history using porosity, density, age, and temperature constraints from IODP Expedition 375. Despite incorporating modern day composition and the temporal evolution of stress and temperature into the pressure solution model, the locations and magnitudes of strain within stylolite clusters observed within Unit IV are not well reproduced at hydrostatic or near-lithostatic fluid pressures. At hydrostatic fluid pressures, we nd agreement at grain sizes corresponding to coarse fossils within the sediment, possibly consistent with porosity loss inferred from stylolite microstructure. At near-lithostatic fluid pressure, observed and modeled strains agree at more common grain sizes in the sample, though this requires sustained periods of high pore fluid during the \sim 60-Myr history of Unit IV. Strain from localized dissolution on stylolites is likely overestimated using a bulk pressure solution model due to the complexity and evolution of subgrain-scale fluid pathways and diffusivities not fully captured within the applied model. We show that a linear scaling of calcite volume percent and strain rate is overly simplistic and that the relationship of grain-scale effects within pressure solution models must be better constrained with composition, time, and strain. We extend the model to the toe of the megathrust and show that dissolution causes negligible change in strain between present day and initial subduction of the sediment column at Site U1520.

Shear within the carbonaceous Unit IV at the plate interface is likely to localize in weak horizontal horizons including clay-rich stylolite clusters at the bottom of Unit IV, slipping by frictional-viscous mechanisms with variable frictional stability due to composition heterogeneity within stylolite clusters. Pressure solution of calcite is likely an important viscous mechanism acting on the grain scale throughout the slow-slipping zone and at shallow depths on the plate interface. Indeed, the down-dip end of slow slip events (15 20 km) roughly correlates with where carbonate solubility becomes negligible. Throughout the slow-slipping zone and the shallow plate interface, other deformation mechanisms in the chalks and clay-rich muds are likely highly variable. The role of carbonate in accommodating shear at the Hikurangi Margin is emphasized, as this dramatically alters slip characteristics and deformation style compared to siliciclastic sediment.

Data Availability Statement

Data from site U1520 are available from IODP (www.iodp.org), and fault and stylolite frequency, sample location, photomicrographs, and SEM data are available from BGS repository No. 130545 (DOI: 10.5285/ af50c4d2-fc87-4052-a41b-226d0590c12b). Code is available on GitHub (DOI: 10.5281/zenodo.3698038).

References

Aharonov, E., & Katsman, R. (2009). Interaction between pressure solution and clays in stylolite development: Insights from modeling. *American Journal of Science*, 309(7), 607–632. https://doi.org/10.2475/07.2009.04

Antriasian, A., Harris, R. N., Tréhu, A. M., Henrys, S. A., Phrampus, B. J., Lauer, R., & Barker, D. (2018). Thermal regime of the Northern Hikurangi Margin, New Zealand. *Geophysical Journal International*, 216(2), 1177–1190. https://doi.org/10.1093/gji/ggy450

Barker, D. H. N., Henrys, S., Caratori Tontini, F., Barnes, P. M., Bassett, D., Todd, E., & Wallace, L. (2018). Geophysical constraints on the relationship between seamount subduction, slow slip, and tremor at the North Hikurangi subduction zone, New Zealand. *Geophysical Research Letters*, 45, 12,804 12,813. https://doi.org/10.1029/2018GL080259

Barker, D. H. N., Sutherland, R., Henrys, S., & Bannister, S. (2009). Geometry of the Hikurangi subduction thrust and upper plate, North Island, New Zealand. *Geochemistry, Geophysics, Geosystems, 10*, Q02007. https://doi.org/10.1029/2008GC002153

Barnes, P. M., Wallace, L. M., Saffer, D. M., Bell, R. E., Underwood, M. B., Fagereng, A., & LeVay, L. J. (2020). Slow slip source characterized by lithological and geometric heterogeneity. *Science Advances*, 6(13), eaay3314. https://doi.org/10.1126/sciadv.aay3314

Baud, P., Rolland, A., Heap, M., Xu, T., Nicolé, M., Ferrand, T., & Conil, N. (2016). Impact of stylolites on the mechanical strength of limestone. *Tectonophysics*, 690, 4 20. https://doi.org/10.1016/j.tecto.2016.03.004

Acknowledgments

This research used samples and data provided by the International Ocean Discovery Program (IODP). Figure 1a was made with Generic Mapping Tools (Wessel et al., 2013). H.L and Å.F. are supported by European Research Council Starting Grant 715836 (•MICA) and NERC IODP Moratorium Grant NE/S002731/1. R.B. is supported by NERC Grant NE/S00291X/1. Funding for F.M. s participation in the Expedition was provided by MIUR (Italian Ministry of Education) for ECORD-IODP-Italia. We thank Duncan Muir for assistance with scanning electron microscopy, Tony Oldroyd for sample preparation, the staff onboard the RV JOIDES Resolution during IODP Expeditions 372 and 375 for their support, and J.P. Gratier, B. Verberne, and A.R. Niemeijer for thorough and constructive reviews which signi cantly improved the manuscript.



- Bell, R., Sutherland, R., Barker, D. H. N., Henrys, S., Bannister, S., Wallace, L., & Beavan, J. (2010). Seismic reflection character of the Hikurangi subduction interface, New Zealand, in the region of repeated Gisborne slow slip events. *Geophysical Journal International*, 180(1), 34–48. https://doi.org/10.1111/j.1365-246X.2009.04401.x
- Bos, B., & Spiers, C. J. (2002). Frictional-viscous flow of phyllosilicate-bearing fault rock: Microphysical model and implications for crustal strength pro les. Journal of Geophysical Research, 107(B2), 2028. https://doi.org/10.1029/2001JB000301
- Boulton, C., Niemeijer, A. R., Hollis, C. J., Townend, J., Raven, M. D., Kulhanek, D. K., & Shepherd, C. L. (2019). Temperature-dependent frictional properties of heterogeneous Hikurangi subduction zone input sediments, ODP Site 1124. *Tectonophysics*, 757, 123 139. https:// doi.org/10.1016/j.tecto.2019.02.006
- Clark, K., Howarth, J., Litch eld, N., Cochran, U., Turnbull, J., Dowling, L., & Wolfe, F. (2019). Geological evidence for past large earthquakes and tsunamis along the Hikurangi subduction margin, New Zealand. *Marine Geology*, 412, 139–172. https://doi.org/10.1016/j. margeo.2019.03.004
- Cox, D. R., & Lewis, P. A. W. (1966). The statistical analysis of series of events, The Statistical Analysis of Series of Events. London: Chapman and Hall.
- Croizé, D., Renard, F., Bjørlykke, K., & Dysthe, D. K. (2010). Experimental calcite dissolution under stress: Evolution of grain contact microstructure during pressure solution creep. Journal of Geophysical Research, 115, B09207. https://doi.org/10.1029/2010JB000869

Davy, B., Hoernle, K., & Werner, R. (2008). Hikurangi Plateau: Crustal structure, rifted formation, and Gondwana subduction history. Geochemistry, Geophysics, Geosystems, 9, Q07004. https://doi.org/10.1029/2007GC001855

- Delahaye, E., Townend, J., Reyners, M., & Rogers, G. (2009). Microseismicity but no tremor accompanying slow slip in the Hikurangi subduction zone, New Zealand. Earth and Planetary Science Letters, 277(1-2), 21 28. https://doi.org/10.1016/j.epsl.2008.09.038
- Delle Piane, C., Clennell, M. B., Keller, J. V., Giwelli, A., & Luzin, V. (2017). Carbonate hosted fault rocks: A review of structural and microstructural characteristic with implications for seismicity in the upper crust. *Journal of Structural Geology*, *103*, 17–36. https://doi. org/10.1016/j.jsg.2017.09.003
- den Hartog, S. A. M., & Spiers, C. J. (2014). A microphysical model for fault gouge friction applied to subduction megathrusts. Journal of Geophysical Research: Solid Earth, 119, 1510 1529. https://doi.org/10.1002/2013JB010580
- Dewers, T., & Ortoleva, P. (1990). Differentiated structures arising from mechano-chemical feedback in stressed rocks. Earth Science Reviews, 29(1-4), 283 298. https://doi.org/10.1016/0012-8252(0)90043-U
- Doser, D. I., & Webb, T. H. (2003). Source parameters of large historical (1917) 1961) earthquakes, North Island, New Zealand. *Geophysical Journal International*, 152, 795) 832.
- Durney, D. W. (1972). Solution-transfer, an important geological deformation mechanism. *Nature*, 235(5337), 315–317. https://doi.org/ 10.1038/235315a0
- Ebner, M., Piazolo, S., Renard, F., & Koehn, D. (2010). Stylolite interfaces and surrounding matrix material: Nature and role of heterogeneities in roughness and microstructural development. *Journal of Structural Geology*, 32(8), 1070–1084. https://doi.org/10.1016/j. jsg.2010.06.014
- Fagereng, Å. (2011a). Fractal vein distributions within a fault-fracture mesh in an exhumed accretionary mélange, Chrystalls Beach Complex, New Zealand. Journal of Structural Geology, 33(5), 918–927. https://doi.org/10.1016/j.jsg.2011.02.009
- Fagereng, Å. (2011b). Wedge geometry, mechanical strength, and interseismic coupling of the Hikurangi subduction thrust, New Zealand. Tectonophysics, 507(1-4), 26 30. https://doi.org/10.1016/j.tecto.2011.05.004
- Fagereng, Å., & Biggs, J. (2019). New perspectives on geological strain rates' calculated from both naturally deformed and actively deforming rocks. Journal of Structural Geology, 125(October 2018), 100 110. https://doi.org/10.1016/j.jsg.2018.10.004
- Ferraro, F., Grieco, D. S., Agosta, F., & Prosser, G. (2018). Space-time evolution of cataclasis in carbonate fault zones. *Journal of Structural Geology*, *110*(June 2017), 45 64. https://doi.org/10.1016/j.jsg.2018.02.007
- Fletcher, R. C., & Pollard, D. D. (1981). Anticrack model for pressure solution surfaces. Geology, 9(9), 419–424. https://doi.org/10.1130/ 0091-7613(1981)9<419:AMFPSS>2.0.CO;2
- Gillespie, P. A., Johnston, J. D., Loriga, M. A., McCaffrey, K. J. W., Walsh, J. J., & Watterson, J. (1999). Influence of layering on vein systematics in line samples. *Geological Society, London, Special Publications, 155*(1), 35–56. https://doi.org/10.1144/GSL.SP.1999.155.01.05
 Grant, J. A. (1986). The isocon diagram-a simple solution to Gresens' equation for metasomatic alteration. *Economic Geology, 81*(8),
- 1976 1982. https://doi.org/10.2113/gsecongeo.81.8.1976 Gratier, J. P., Dysthe, D. K., & Renard, F. (2013). The role of pressure solution creep in the ductility of the Earth's upper crust. *Advances in Geophysics*, *54*, 47 179. https://doi.org/10.1016/B978-0-12-380940-7.00002-0

Gratier, J. P., Noiriel, C., & Renard, F. (2015). Experimental evidence for rock layering development by pressure solution. *Geology*, 43(10), 871 874. https://doi.org/10.1130/G36713.1

- Gratier, J. P., Renard, F., & Labaume, P. (1999). How pressure solution creep and fracturing processes interact in the upper crust to make it behave in both a brittle and viscous manner. *Journal of Structural Geology*, 21(8-9), 1189 1197. https://doi.org/10.1016/S0191-8141(99)00035-8
- Greene, G. W., Kristiansen, K., Meyer, E. E., Boles, J. R., & Israelachvili, J. N. (2009). Role of electrochemical reactions in pressure solution. *Geochimica et Cosmochimica Acta*, 73(10), 2862–2874. https://doi.org/10.1016/j.gca.2009.02.012

Gundersen, E., Renard, F., Dysthe, D. K., Bjørlykke, K., & Jamtveit, B. (2002). Coupling between pressure solution creep and diffusive mass transport in porous rocks. *Journal of Geophysical Research*, *107*(B11), ECV 19 1 ECV 19 19. https://doi.org/10.1029/2001JB000287
 Heald, M. T. (1955). Stylolites in sandstones. *The Journal of Geology*, *63*(2), 101 114. https://doi.org/10.1086/626237

Heap, M., Reuschlé, T., Baud, P., Renard, F., & Iezzi, G. (2018). The permeability of stylolite-bearing limestone. Journal of Structural Geology, 116(May), 81 93. https://doi.org/10.1016/j.jsg.2018.08.007

Heilbronner, R., & Barrett, S. (2014). Image analysis in Earth sciences. Berlin, Heidelberg: Springer Berlin Heidelberg. https://doi.org/ 10.1007/978-3-642-10343-8

Hickman, S. H., & Evans, B. (1995). Kinetics of pressure solution at halite-silica interfaces and intergranular clay lms. Journal of Geophysical Research, 100(B7), 13,113 13,132. https://doi.org/10.1029/95JB00911

- Ikari, M. J., Niemeijer, A. R., Spiers, C. J., Kopf, A. J., & Saffer, D. M. (2013). Experimental evidence linking slip instability with seafloor lithology and topography at the Costa Rica convergent margin. *Geology*, 41(8), 891–894. https://doi.org/10.1130/G33956.1
- Imber, J., Holdsworth, R. E., Smith, S. A. F., Jefferies, S. P., & Collettini, C. (2008). Frictional-viscous flow, seismicity and the geology of weak faults: A review and future directions. *Geological Society, London, Special Publications*, 299(1), 151–173. https://doi.org/10.1144/ SP299.10
- Kawabata, K., Tanaka, H., & Kimura, G. (2007). Mass transfer and pressure solution in deformed shale of accretionary complex: Examples from the Shimanto Belt, southwestern Japan. Journal of Structural Geology, 29(4), 697 711. https://doi.org/10.1016/j.jsg.2006.11.009

- Kennedy, L., & White, J. (2001). Low-temperature recrystallization in calcite: Mechanisms and consequences. *Geology*, 29(11), 1027. https://doi.org/10.1130/0091-7613(2001)029<1027:LTRICM>2.0.CO;2
- Koehn, D., Rood, M. P., Beaudoin, N., Chung, P., Bons, P. D., & Gomez-Rivas, E. (2016). A new stylolite classi cation scheme to estimate compaction and local permeability variations. *Sedimentary Geology*, 346, 60 71. https://doi.org/10.1016/j.sedgeo.2016.10.007
- Kurzawski, R. M., Niemeijer, A. R., Stipp, M., Charpentier, D., Behrmann, J. H., & Spiers, C. J. (2018). Frictional properties of subduction input sediments at an erosive convergent continental margin and related controls on décollement slip modes: The Costa Rica seismogenesis project. Journal of Geophysical Research: Solid Earth, 123, 8385–8408. https://doi.org/10.1029/2017JB015398
- Lacroix, B., Tesei, T., Oliot, E., Lah d, A., & Collettini, C. (2015). Early weakening processes inside thrust fault. *Tectonics*, 34(7), 1396 1411. https://doi.org/10.1002/2014TC003716
- Laronne Ben-Itzhak, L., Aharonov, E., Karcz, Z., Kaduri, M., & Toussaint, R. (2014). Sedimentary stylolite networks and connectivity in limestone: Large-scale eld observations and implications for structure evolution. *Journal of Structural Geology*, 63, 106–123. https://doi. org/10.1016/j.jsg.2014.02.010
- Leah, H., Fondriest, M., Lucca, A., Storti, F., Balsamo, F., & Di Toro, G. (2018). Coseismic extension recorded within the damage zone of the Vado di Ferruccio Thrust Fault, Central Apennines, Italy. *Journal of Structural Geology*, 114, 121 138. https://doi.org/10.1016/j. jsg.2018.06.015
- Lehner, F. K. (1995). A model for intergranular pressure solution in open systems. *Tectonophysics*, 245(3-4), 153 170. https://doi.org/ 10.1016/0040-1951(94)00232-X
- Lind, I. (1993). Stylolites in Chalk from Leg 130 Ontong Java Plateau. Proceedings of the Ocean Drilling Program, Scientific Results, 130, 445 451.
- Lyakhovsky, V., Ben-Zion, Y., & Agnon, A. (1997). Distributed damage, faulting, and friction. Journal of Geophysical Research, 102(B12), 27,635 27,649. https://doi.org/10.1029/97JB01896
- Macente, A., Fusseis, F., Butler, I. B., Tudisco, E., Hall, S. A., & Andò, E. (2018). 4D porosity evolution during pressure-solution of NaCl in the presence of phyllosilicates. *Earth and Planetary Science Letters*, 502, 115–125. https://doi.org/10.1016/j.epsl.2018.08.032
- Mitchell, T. M., & Faulkner, D. R. (2009). The nature and origin of off-fault damage surrounding strike-slip fault zones with a wide range of displacements: A eld study from the Atacama fault system, northern Chile. Journal of Structural Geology, 31(8), 802 816. https://doi. org/10.1016/j.jsg.2009.05.002
- Mittempergher, S., Cerchiari, A., Francesca, R., & Festa, A. (2018). From soft sediment deformation to fluid assisted faulting in the shallow part of a subduction megathrust analogue: The Sestola Vidiciatico tectonic unit (Northern Apennines, Italy). *Geological Magazine*, 155(2), 438–450. https://doi.org/10.1017/S0016756817000668
- Moore, J., & Mascle, A. (1990). Proceedings of the Ocean Drilling Program, 110 Scienti c Results. (Vol. 110), Ocean Drilling Program, https://doi.org/10.2973/odp.proc.sr.110.1990
- Morris, J., Villinger, H., & Klaus, A. (2006). Proceedings of the Ocean Drilling Program, 205 Scienti c Results. (Vol. 205), Ocean Drilling Program, https://doi.org/10.2973/odp.proc.sr.205.2006
- Nakashima, S. (1995). Diffusivity of ions in pore water as a quantitative basis for rock deformation rate estimates. *Tectonophysics*, 245(3-4), 185 203. https://doi.org/10.1016/0040-1951(94)00234-Z
- Nayar, K. G., Sharqawy, M. H., Banchik, L. D., & Lienhard, J. H. (2016). Thermophysical properties of seawater: A review and new correlations that include pressure dependence. *Desalination*, 390, 1 24. https://doi.org/10.1016/j.desal.2016.02.024
- Niemeijer, A., Spiers, C. J., & Bos, B. (2002). Compaction creep of quartz sand at 400-600°C: Experimental evidence for
- dissolution-controlled pressure solution. Earth and Planetary Science Letters, 195(3-4), 261 275. https://doi.org/10.1016/S0012-821X(01)00593-3
- Otsu, N. (1979). A threshold selection method from gray-level histograms. *IEEE Transactions on Systems, Man, and Cybernetics, 9*(1), 62 66. https://doi.org/10.1109/TSMC.1979.4310076
- Paterson, M. S., & Wong, T. F. (2005). Experimental rock deformation---The brittle field Edited by Second, Co. Berlin/Heidelberg: Springer-Verlag. https://doi.org/10.1007/b137431
- Peacock, D. C. P., & Azzam, I. N. (2006). Development and scaling relationships of a stylolite population. Journal of Structural Geology, 28(10), 1883–1889. https://doi.org/10.1016/j.jsg.2006.04.008
- P ffner, O. a., & Ramsay, J. G. (1982). Constraints on geological strain rates: Arguments from nite strain states of naturally deformed rocks. *Journal of Geophysical Research*, 87(B1), 311 321. https://doi.org/10.1029/JB087iB01p00311
- Plummer, L., & Busenberg, E. (1982). The solubilities of calcite, aragonite and vaterite in CO2-H2O solutions between 0 and 90°C, and an evaluation of the aqueous model for the system CaCO3-CO2-H2O. *Geochimica et Cosmochimica Acta*, 46(6), 1011 1040. https://doi.org/ 10.1016/0016-7037(82)90056-4
- Pluymakers, A. M. H., & Spiers, C. J. (2015). Compaction creep of simulated anhydrite fault gouge by pressure solution: theory v. experiments and implications for fault sealing. *Geological Society, London, Special Publications*, 409(1), 107–124. https://doi.org/10.1144/ SP409.6
- Rabinowitz, H. S., Savage, H. M., Skarbek, R. M., Ikari, M. J., Carpenter, B. M., & Collettini, C. (2018). Frictional behavior of input sediments to the Hikurangi Trench, New Zealand. *Geochemistry, Geophysics, Geosystems, 19*, 2973–2990. https://doi.org/10.1029/ 2018GC007633
- Railsback, L. B. (1993). Lithologic controls on morphology of pressure-dissolution surfaces (stylolites and dissolution seams) in Paleozoic carbonate rocks from the mideastern United States. *Journal of Sedimentary Research*, 63(3), 513–522. https://doi.org/ 10.2110/jsr.63.513
- Railsback, L. B. (1998). Evaluation of spacing of stylolites and its implication for self-organisation of pressure dissolution. Journal of Sedimentary Research, 68(1), 2 7. https://doi.org/10.2110/jsr.68.2
- Raj, R. (1982). Creep in polycrystalline aggregates by matter transport through a liquid phase. *Journal of Geophysical Research*, 87(B6), 4731 4739. https://doi.org/10.1029/JB087iB06p04731
- Renard, F., Gratier, J. P., & Jamtveit, B. (2000). Kinetics of crack-sealing, intergranular pressure solution, and compaction around active faults. *Journal of Structural Geology*, 22(10), 1395–1407. https://doi.org/10.1016/S0191-8141(00)00064-X
- Renard, F., Ortoleva, P., & Gratier, J. P. (1997). Pressure solution in sandstones: Influence of clays and dependence on temperature and stress. *Tectonophysics*, 280(3-4), 257 266. https://doi.org/10.1016/S0040-1951(97)00039-5
- Rutter, E. H. (1976). The kinetics of rock deformation by pressure solution [and discussion]. *Philosophical Transactions of the Royal Society* A: Mathematical, Physical and Engineering Sciences, 283(1312), 203 219. https://doi.org/10.1098/rsta.1976.0079
- Rutter, E. H. (1983). Pressure solution in nature, theory and experiment. Journal of the Geological Society, 140(5), 725 740. https://doi.org/ 10.1144/gsjgs.140.5.0725

Rutter, E. H., Hackston, A. J., Yeatman, E., Brodie, K. H., Mecklenburgh, J., & May, S. E. (2013). Reduction of friction on geological faults by weak-phase smearing. *Journal of Structural Geology*, *51*, 52 60. https://doi.org/10.1016/j.jsg.2013.03.008

Safaricz, M., & Davison, I. (2005). Pressure solution in chalk. AAPG Bulletin, 89(3), 383 401. https://doi.org/10.1306/10250404015

Sample, J. C. (1990). The effect of carbonate cementation of underthrust sediments on deformation styles during underplating. Journal of Geophysical Research, 95(B6), 9111. https://doi.org/10.1029/JB095iB06p09111

- Skarbek, R. M., & Rempel, A. W. (2017). Heterogeneous Coulomb wedges: Influence of fluid pressure, porosity, and application to the Hikurangi subduction margin, New Zealand. *Journal of Geophysical Research: Solid Earth*, 122, 1585–1613. https://doi.org/10.1002/ 2016JB013497
- Spiers, C. J., Meer de, S., Niemeijer, A, & Zhang, X. (2004). Kinetics of rock deformation by pressure solution and the role of thin aqueous lms.pdf. In S. Nakashima, C. J. Spiers, L. Mercury, P. A. Fenter, & M. F. Hochella Jr. (Eds.), *Physicochemistry of water in geological and biological systems* (pp. 129–158). Tokyo, Japan: Universal Academy Press.
- Spiers, C. J., Schutjens, P. M., Brzesowsky, R. H., Peach, C. J., Liezenberg, J. L., & Zwart, H. J. (1990). Experimental determination of constitutive parameters governing creep of rocksalt by pressure solution. *Geological Society Special Publication*, 54(54), 215–227. https:// doi.org/10.1144/GSL.SP.1990.054.01.21

Tada, R., & Siever, R. (1989). Pressure solution during diagenesis. Annual Review Earth and Planetary Sciences, 17(1), 89 118. https://doi. org/10.1038/264511b0

- Tesei, T., Collettini, C., Barchi, M. R., Carpenter, B. M., & Di Stefano, G. (2014). Heterogeneous strength and fault zone complexity of carbonate-bearing thrusts with possible implications for seismicity. *Earth and Planetary Science Letters*, 408, 307–318. https://doi.org/ 10.1016/j.epsl.2014.10.021
- Todd, E. K., & Schwartz, S. Y. (2016). Tectonic tremor along the northern Hikurangi Margin, New Zealand, between 2010 and 2015. Journal of Geophysical Research: Solid Earth, 121, 8706–8719. https://doi.org/10.1002/2016JB013480
- Todd, E. K., Schwartz, S. Y., Mochizuki, K., Wallace, L. M., Sheehan, A. F., Webb, S. C., & Ito, Y. (2018). Earthquakes and tremor linked to seamount subduction during shallow slow slip at the Hikurangi Margin, New Zealand. *Journal of Geophysical Research: Solid Earth*, 123, 6769–6783. https://doi.org/10.1029/2018JB016136
- Tondi, E., Antonellini, M., Aydin, A., Marchegiani, L., & Cello, G. (2006). The role of deformation bands, stylolites and sheared stylolites in fault development in carbonate grainstones of Majella Mountain, Italy. *Journal of Structural Geology*, *28*(3), 376–391. https://doi.org/10.1016/j.jsg.2005.12.001
- Toussaint, R., Aharonov, E., Koehn, D., Gratier, J. P., Ebner, M., Baud, P., & Renard, F. (2018). Stylolites: A review. Journal of Structural Geology, 114(May), 163 195. https://doi.org/10.1016/j.jsg.2018.05.003
- van den Ende, M. P., Niemeijer, A. R., & Spiers, C. J. (2019). Influence of grain boundary structural evolution on pressure solution creep rates. *Journal of Geophysical Research: Solid Earth*, 124, 10,210 10,230. https://doi.org/10.1029/2019JB017500
- van-den-Ende, M. P., Marketos, G., Niemeijer, A. R., & Spiers, C. J. (2018). Investigating compaction by intergranular pressure solution using the discrete element method. *Journal of Geophysical Research: Solid Earth*, 123, 107–124. https://doi.org/10.1002/ 2017JB014440
- Verberne, B. A., de Bresser, J. H., Niemeijer, A. R., Spiers, C. J., de Winter, D. A., & Plümper, O. (2013). Nanocrystalline slip zones in calcite fault gouge show intense crystallographic preferred orientation: Crystal plasticity at subseismic slip rates at 18-150°C. Geology, 41(8), 863 866. https://doi.org/10.1130/G34279.1
- Verberne, B. A., & Spiers, C. J. (2017). A quantitative microstructural investigation of depleted and undepleted reservoir sandstones. In 51st US Rock Mechanics/Geomechanics Symposium 2017, 5, pp. 3396–3405.
- Verberne, B. A., Spiers, C. J., Niemeijer, A. R., De Bresser, J. H. P., De Winter, D. A. M., & Plümper, O. (2014). Frictional properties and microstructure of calcite-rich fault gouges sheared at sub-seismic sliding velocities. *Pure and Applied Geophysics*, 171(10), 2617–2640. https://doi.org/10.1007/s00024-013-0760-0
- Villinger, H., Grevemeyer, I., Kaul, N., Hauschild, J., & Pfender, M. (2002). Hydrothermal heat flux through aged oceanic crust: Where does the heat escape? *Earth and Planetary Science Letters*, 202(1), 159–170. https://doi.org/10.1016/S0012-821X(02)00759-8
- Viti, C., Collettini, C., & Tesei, T. (2014). Pressure solution seams in carbonatic fault rocks: mineralogy, micro/nanostructures and deformation mechanism. Contributions to Mineralogy and Petrology, 167(2), 970. https://doi.org/10.1007/s00410-014-0970-1
- Wallace, L. M. (2004). Subduction zone coupling and tectonic block rotations in the North Island, New Zealand. Journal of Geophysical Research, 109, B12406. https://doi.org/10.1029/2004JB003241
- Wallace, L. M., & Beavan, J. (2010). Diverse slow slip behavior at the Hikurangi subduction margin, New Zealand. Journal of Geophysical Research, 115, B12402. https://doi.org/10.1029/2010JB007717
- Wallace, L. M., Cochran, U., Power, W., & Clark, K. (2014). Earthquake and tsunami potential of the Hikurangi subduction thrust, New Zealand: Insights from paleoseismology, GPS, and tsunami modeling. Oceanography, 27(2), 104 117. https://doi.org/10.5670/ oceanog.2014.46
- Wallace, L. M., Saffer, D. M., Barnes, P. M., Pecher, I. A., Petronotis, K. E., LeVay, L. J., & the Expedition 372/375 (2019). Hikurangi subduction margin coring, logging, and observatories. 375, College Station, TX, International Ocean Discovery Program, https://doi.org/ 10.14379/iodp.proc.372B375.2019
- Wallace, L. M., Webb, S. C., Ito, Y., Mochizuki, K., Hino, R., Henrys, S., & Sheehan, A. F. (2016). Slow slip near the trench at the Hikurangi subduction zone, New Zealand. Science, 352(6286), 701 704. https://doi.org/10.1126/science.aaf2349
- Watkinson, A. J., & Ward, E. G. (2006). Reactivation of pressure-solution seams by a strike-slip fault-sequential, dilational jog formation and fluid flow. AAPG Bulletin, 90(8), 1187 1200. https://doi.org/10.1306/03060604088
- Wessel, P., Smith, W. H., Scharroo, R., Luis, J., & Wobbe, F. (2013). Generic mapping tools: Improved version released. *Eos*, 94(45), 409–410. https://doi.org/10.1002/2013EO450001
- Willemse, E. J., Peacock, D. C., & Aydin, A. (1997). Nucleation and growth of strike-slip faults in limestones from Somerset, U.K. Journal of Structural Geology, 19(12), 1461 1477. https://doi.org/10.1016/S0191-8141(97)00056-4
- Yabe, S., Ide, S., & Yoshioka, S. (2014). Along-strike variations in temperature and tectonic tremor activity along the Hikurangi subduction zone, New Zealand. *Earth, Planets and Space*, 66(1), 142. https://doi.org/10.1186/s40623-014-0142-6
- Zhang, X., Salemans, J., Peach, C. J., & Spiers, C. J. (2002). Compaction experiments on wet calcite powder at room temperature: Evidence for operation of intergranular pressure solution. *Geological Society, London, Special Publications*, 200(1), 29–39. https://doi.org/10.1144/ GSL.SP.2001.200.01.02

Shaddox, H. R., & Schwartz, S. Y. (2019). Subducted seamount diverts shallow slow slip to the forearc of the northern Hikurangi subduction zone, New Zealand. Geology, 47(5), 415–418. https://doi.org/10.1130/G45810.1



- Zhang, X., & Spiers, C. J. (2005). Compaction of granular calcite by pressure solution at room temperature and effects of pore fluid chemistry. *International Journal of Rock Mechanics and Mining Sciences*, 42(7-8 SPEC. ISS.), 950–960. https://doi.org/10.1016/j. ijrmms.2005.05.017
- Zhang, X., Spiers, C. J., & Peach, C. J. (2010). Compaction creep of wet granular calcite by pressure solution at 28°C to 150°C. Journal of Geophysical Research, 115, B09217. https://doi.org/10.1029/2008JB005853
- Zubtsov, S., Renard, F., Gratier, J. P., Guiguet, R., Dysthe, D. K., & Traskine, V. (2004). Experimental pressure solution compaction of synthetic halite/calcite aggregates. *Tectonophysics*, 385(1-4), 45 57. https://doi.org/10.1016/j.tecto.2004.04.016